# **INDENTURE**

by and among

# POMONA PUBLIC FINANCING AUTHORITY

and

**CITY OF POMONA** 

and

ZIONS BANK, a division of ZB, N.A., as Trustee

Dated as of \_\_\_\_\_\_1, 2017

Relating to

\$\_\_\_\_\_POMONA PUBLIC FINANCING AUTHORITY
2017 LEASE REVENUE REFUNDING BONDS, SERIES BG
(FEDERALLY TAXABLE)

# TABLE OF CONTENTS

		Page
	ARTICLE I	
	DEFINITIONS; EQUAL SECURITY	
1.01	Definitions	2
1.02	Equal Security	
	A DITICLE II	
	ARTICLE II THE BONDS	
2.01	Authorization of Bonds	12
2.01	Terms of Series BG Bonds	
2.02	Form of Series BG Bonds	
2.03	Transfer and Exchange of Bonds	
2.05	Registration Books	
2.06	Execution of Bonds	
2.07	Authentication of Bonds	
2.08	Temporary Bonds	
2.09	Bonds Mutilated, Lost, Destroyed or Stolen	
2.10	Book-Entry Bonds	17
	ARTICLE III	
	ISSUANCE OF BONDS; APPLICATION OF PROCEEDS	
3.01	Issuance of Series BG Bonds	18
3.02	Application of Proceeds of the Series BG Bonds	
3.03	Costs of Issuance Fund	
3.04	[Reserved.]	
3.05	Conditions for the Issuance of Additional Bonds	19
3.06	Procedure for the Issuance of Additional Bonds	20
3.07	Additional Bonds	21
	ARTICLE IV	
	REDEMPTION OF BONDS	
4.01	Redemption of Series BG Bonds	21
4.02	Notice of Redemption	22
4.03	Selection of Bonds for Redemption	
4.04	Partial Redemption of Bonds	
4.05	Effect of Notice of Redemption	23
	ARTICLE V	
	SECURITY FOR BONDS; FLOW OF FUNDS; INVESTMENTS	
5.01	Pledge; Special Obligations	
5.02	Flow of Funds	
5.03	Application of Net Insurance Proceeds	
5.04	Title Insurance	
5.05	Reserve Fund	
5.06	[Reserved.]	
5.07	Investment of Moneys	29

# TABLE OF CONTENTS

(continued)

		Page
5.08	[Claims Upon the Insurance Policy and Payments by and to the Insurer.]	
5.09	Payments by the Insurer as a Result of Nonpayment	30
	ARTICLE VI	
	COVENANTS	
6.01	Compliance with Agreements	30
6.02	Compliance with Ground Lease and Lease Agreement	
6.03	Observance of Laws and Regulations	
6.04	Other Liens	
6.05	Prosecution and Defense of Suits	
6.06	Accounting Records and Statements	
6.07	Recordation and Filing	
6.08	[Reserved]	
6.09	Continuing Disclosure	31
6.10	Further Assurances	32
	ARTICLE VII	
	DEFAULT AND LIMITATIONS OF LIABILITY	
7.01	Action on Default	
7.02	Other Remedies of the Trustee	
7.03	Non-Waiver	
7.04	Remedies Not Exclusive	
7.05	No Liability by the Authority to the Owners	
7.06	No Liability by the City to the Owners	
7.07 7.08	No Liability of the Trustee to the Owners	
7.08	Application of Amounts After Default  Trustee May Enforce Claims Without Possession of Bonds	
7.10	Limitation on Suits	
7.10	Insurer Rights	
	A DITICUE VIII	
	ARTICLE VIII THE TRUSTEE	
8.01	Employment of the Trustee	35
8.02	Duties, Removal and Resignation of the Trustee	35
8.03	Compensation of the Trustee	36
8.04	Protection of the Trustee	
8.05	Notice to Insurer by Trustee	39
	ARTICLE IX	
	MODIFICATION OR AMENDMENTS	
9.01	Modifications and Amendments Permitted	
9.02	Effect of Supplemental Indenture	
9.03	Endorsement of Bonds; Preparation of New Bonds	
9.04	Amendment of Particular Bonds	
9.05	Effect of Insurance Policy	41

# TABLE OF CONTENTS

(continued)

		Page
	ARTICLE X DEFEASANCE	
10.01	Discharge of Indenture	<i>1</i> 1
10.01	Bonds Deemed To Have Been Paid	
10.02	Payment of Bonds After Discharge of Indenture	
10.03	rayment of Bonds After Discharge of Indenture	42
	ARTICLE XI MISCELLANEOUS	
11.01	Benefits of Indenture Limited to Parties	
11.02	Successor Deemed Included in all References to Predecessor	
11.03	Execution of Documents by Owners	
11.04	Waiver of Personal Liability	
11.05	Destruction of Bonds	
11.06		
11.07	Article and Section Headings Gender and References	
11.08	Partial Invalidity	
11.09	Disqualified Bonds	
11.10	Money Held for Particular Bonds	
11.11	Payment on Non-Business Days	
11.12	California Law	
11.13	Notices	
11.14	Notice to Rating Agencies	
11.15	Execution in Counterparts	
11.16		
11.17	Impairment of Insurer's Rights	
11.18	Insurer Consideration	
11.19	Amounts Paid by Insurer	
11.20	Covenant to Preserve Priority	
11.21	Subrogation and Survival of Obligations	
11.22	Reimbursement of Fees	
11.23	Provision of Information	
11.24	Additional Information	
11.25	Discussion of and Access to Information	48
EXHII	BIT A FORM OF SERIES BG BOND	A-1

#### **INDENTURE**

**THIS INDENTURE** (this "Indenture"), executed and entered into as of \_\_\_\_\_\_\_\_1, 2017, is by and among the POMONA PUBLIC FINANCING AUTHORITY, a joint exercise of powers entity duly organized and existing under the laws of the State of California (the "Authority"), the CITY OF POMONA, a municipal corporation and charter city duly organized and existing under and by virtue of the Constitution and laws of the State of California (the "City") and Zions Bank, a division of ZB, N.A., a national banking association duly organized and existing under and by virtue of the laws of the United States, as Trustee (the "Trustee");

#### **WITNESSETH:**

**WHEREAS**, the City desires to defease the Authority's outstanding 2006 Lease Revenue Bonds, Series AU and Series AV and a portion of the City's outstanding Pension Obligation Refunding Bonds (Federally Taxable) Series 2006 AR (collectively, the "Refunded Bonds");

**WHEREAS**, in order to refund the Refunded Bonds, the City will lease certain real property and the improvements located thereon (the "Property") to the Authority pursuant to a Ground Lease, dated as of the date hereof, and the City will sublease the Property back from the Authority pursuant to a Lease Agreement, dated as of the date hereof (the "Lease Agreement");

**WHEREAS**, the City and the Authority have determined that it would be in the best interests of the City and the Authority to provide the funds necessary to refund the Refunded Bonds through the issuance by the Authority of bonds payable from the base rental payments (the "Base Rental Payments") to be made by the City under the Lease Agreement;

**WHEREAS**, all rights to receive the Base Rental Payments have been assigned without recourse by the Authority to the Trustee pursuant to an Assignment Agreement, dated as of the date hereof:

**WHEREAS**, the Authority and the City desire to provide for the issuance by the Authority of Pomona Public Financing Authority 2017 Lease Revenue Refunding Bonds, Series BG (Federally Taxable) (the "Series BG Bonds"), in the aggregate principal amount of \$\_\_\_\_\_\_, in order to refund the Refunded Bonds;

**WHEREAS**, the Series BG Bonds will be payable equally and ratably from the Base Rental Payments;

**WHEREAS**, the Authority and the City desire to provide for the issuance of additional bonds (the "Additional Bonds") payable from the Base Rental Payments on a parity with the Series BG Bonds (the Series BG Bonds and any such Additional Bonds being collectively referred to as the "Bonds");

WHEREAS, in order to provide for the authentication and delivery of the Bonds, to establish and declare the terms and conditions upon which the Bonds are to be issued and secured and to secure the payment of the principal thereof, premium, if any, and interest thereon, the Authority and the City have authorized the execution and delivery of this Indenture; and

WHEREAS, the Authority and the City have determined that all acts and proceedings required by law necessary to make the Bonds, when executed by the Authority, authenticated and delivered by the Trustee and duly issued, the valid, binding and legal special obligations of the Authority, and to constitute this Indenture a valid and binding agreement for the uses and purposes herein set forth in accordance with its terms, have been done and taken, and the execution and delivery of the Indenture has been in all respects duly authorized;

**NOW THEREFORE**, in consideration of the premises and of the mutual agreements and covenants contained herein and for other valuable consideration, the parties do hereby agree as follows:

#### ARTICLE I

## **DEFINITIONS; EQUAL SECURITY**

- **1.01 Definitions**. Unless the context otherwise requires, the terms defined in this Section shall for all purposes hereof and of any amendment hereof or supplement hereto and of the Bonds and of any certificate, opinion, request or other document mentioned herein or therein have the meanings defined herein, the following definitions to be equally applicable to both the singular and plural forms of any of the terms defined herein. Capitalized terms not otherwise defined herein shall have the meanings assigned to such terms in the Lease Agreement.
- "Additional Bonds" means Bonds other than the Series BG Bonds issued hereunder in accordance with the provisions of Sections 3.06 and 3.07 hereof.
- "Act" means the Marks-Roos Local Bond Pooling Act of 1985, commencing with Section 6584 of the California Government Code.
- "Additional Rental Payments" means all amounts payable by the City as Additional Rental Payments pursuant to Section 3.02 of the Lease Agreement.
- "Assignment Agreement" means the Assignment Agreement, dated as of the date hereof, by and between the Authority and the Trustee.
- "Authority" means the Pomona Public Financing Authority, a joint exercise of powers entity organized and existing under and by virtue of the laws of the State of California.
- "Authorized Authority Representative" means the Chair, the Vice Chair or the Treasurer of the Authority, or any other person authorized by the Board of Directors of the Authority to act on behalf of the Authority under or with respect to this Indenture.
- "Authorized City Representative" means the Mayor of the City, the City Manager of the City, the Assistant City Manager of the City, the Finance Director of the City, or any other person authorized by the City Council of the City to act on behalf of the City under or with respect to this Indenture.
  - "Authorized Denominations" means \$5,000 or any integral multiple thereof.
- "Base Rental Payment Fund" means the fund by that name established in accordance with Section 5.02 hereof.

- "Base Rental Payments" means all amounts payable to the Authority by the City as Base Rental Payments pursuant to Section 3.01 of the Lease Agreement.
- "Beneficial Owner" means, whenever used with respect to a Book-Entry Bond, the person whose name is recorded as the beneficial owner of such Book-Entry Bond or a portion of such Book-Entry Bond by a Participant on the records of such Participant or such person's subrogee.
  - "Bonds" means the Series BG Bonds and any Additional Bonds issued hereunder.
- "Book-Entry Bonds" means the Bonds of a Series registered in the name of the nominee of DTC, or any successor securities depository for such Series of Bonds, as the registered owner thereof pursuant to the terms and provisions of Section 2.10 hereof.
- "Business Day" means a day which is not (a) a Saturday, Sunday or legal holiday, (b) a day on which banking institutions in the State of California, or in any state in which the Office of the Trustee is located, are required or authorized by law (including executive order) to close, or (c) a day on which the New York Stock Exchange is closed.
- "Cede & Co." means Cede & Co., the nominee of DTC, and any successor nominee of DTC with respect to a Series of Book-Entry Bonds.
- "City" means the City of Pomona, a municipal corporation and charter city duly organized and existing under and by virtue of the Constitution and laws of the State of California.

"Closing Date"	means	, 201	7
----------------	-------	-------	---

"Code" means the Internal Revenue Code of 1986.

- "Continuing Disclosure Certificate" means the Continuing Disclosure Certificate, dated as of the date hereof, executed by the City, as originally executed and as it may from time to time be amended in accordance with the provisions thereof.
- "Costs of Issuance" means all the costs of issuing and delivering the Bonds, including, but not limited to, all printing and document preparation expenses in connection with this Indenture, the Lease Agreement, the Ground Lease, the Assignment Agreement, the Bonds and any preliminary official statement and final official statement pertaining to the Bonds, rating agency fees, CUSIP Service Bureau charges, market study fees, legal fees and expenses of counsel with the issuance and delivery of the Bonds, the initial fees and expenses of the Trustee and Escrow Agent and its counsel and other fees and expenses incurred in connection with the issuance and delivery of the Bonds, to the extent such fees and expenses are approved by the City.
- "Costs of Issuance Fund" means the fund by that name established in accordance with Section 3.04 hereof.
- "DTC" means The Depository Trust Company, a limited-purpose trust company organized under the laws of the State of New York, and its successors as securities depository for any Series of Book-Entry Bonds, including any such successor appointed pursuant to Section 2.10 hereof.
- "Escrow Agreement (Series AR)" means that certain Escrow Agreement dated as of \_\_\_\_\_\_\_1, 2017, by and between the City and the Escrow Bank, providing for the defeasance of

the City's outstanding Pension Obligation Refunding Bonds (Federally Taxable) Series 2006 AR, that were issued as current interest bonds.

"Escrow Agreement (Series AU)" means that certain Escrow Agreement dated as of \_\_\_\_\_\_1, 2017, by and among the City, the Authority and the Escrow Bank, providing for the defeasance of the Authority's outstanding 2006 Lease Revenue Bonds, Series AU.

"Escrow Agreement (Series AV)" means that certain Escrow Agreement dated as of \_\_\_\_\_\_1, 2017, by and among the City, the Authority and the Escrow Bank, providing for the defeasance of the Authority's outstanding 2006 Lease Revenue Bonds, Series AV.

"Escrow Bank" means The Bank of New York Mellon Trust Company, N.A.

"Series AR Escrow Fund" means the Escrow Fund established and held by the Escrow Bank pursuant to the Escrow Agreement (Series AR).

"Series AU Escrow Fund" means the Escrow Fund established and held by the Escrow Bank pursuant to the Escrow Agreement (Series AU).

"Series AV Escrow Fund" means the Escrow Fund established and held by the Escrow Bank pursuant to the Escrow Agreement (Series AV).

"Federal Securities" means (1) cash, (2) non-callable direct obligations of the United States of America ("Treasuries"), (3) evidences of ownership of proportionate interests in future interest and principal payments on Treasuries held by a bank or trust company as custodian, under which the owner of the investment is the real party in interest and has the right to proceed directly and individually against the obligor and the underlying Treasuries are not available to any person claiming through the custodian or to whom the custodian may be obligated, (4) subject to the prior written consent of the Insurer (so long as the Insurer is not in default on any obligation under the Insurance Policy or the Series BG Reserve Surety Policy), pre-refunded municipal obligations rated "AAA" and "Aaa" by S&P and Moody's, respectively, or (5) subject to the prior written consent of the Insurer (so long as the Insurer is not in default on any obligation under the Insurance Policy or the Series BG Reserve Surety Policy), securities eligible for "AAA" defeasance under then existing criteria of S&P or any combination thereof, shall be used to effect defeasance of the Series BG Bonds unless the Insurer otherwise approves.

"Forward Purchase and Sale Agreement" means an agreement entered into by the Trustee and/or the Authority and/or the City and a bank or financial institution (the "Provider") rated "A2" or higher by Moody's and "A" or higher by S&P providing for the Provider to tender, and for the Trustee to purchase, certain eligible securities on one or more dates occurring at least thirty (30) business days after the date of such agreement; provided that (1) securities tendered by the Provider are purchased on a delivery versus payment basis, (2) securities purchased constitute Permitted Investments at the time they are tendered, and (3) the Authority and the City receive an opinion of counsel acceptable to the Authority, to the City and to the Trustee which states that the agreement constitutes a legally valid, binding, and enforceable obligation of the Provider and that in the event of a bankruptcy of the Provider, securities sold by the Provider to the Trustee pursuant to the agreement do not constitute property of the estate of the Provider within the applicable bankruptcy or insolvency laws.

"Ground Lease" means the Ground Lease, dated as of the date hereof, by and between the City and the Authority, as originally executed and as it may from time to time be amended in accordance with the provisions thereof and of the Lease Agreement.

"**Indenture**" means this Indenture, as originally executed and as it may be amended or supplemented from time to time by any Supplemental Indenture.

"Information Services" means Municipal Securities Rulemaking Board through the Electronic Municipal Marketplace Access (EMMA) website; and, in accordance with then current guidelines of the Securities and Exchange Commission, such other services providing information with respect to called bonds as the Authority may designate in a Written Certificate of the Authority delivered to the Trustee.

"Insurance Policy" means the insurance policy issued by the Insurer guaranteeing the scheduled payment of principal of and interest on the Series BG Bonds when due.

"Insurer"	means	, or any	successor	thereto	or assignee	thereof	f.

"**Interest Fund**" means the fund by that name established in accordance with Section 5.02 hereof.

"Interest Payment Date" means January 1 and July 1 of each year, commencing on [July 1, 2017].

"Lease Agreement" means the Lease Agreement, dated as of the date hereof, by and between the City and the Authority, as originally executed and as it may be from time to time amended in accordance with the provisions thereof.

"Moody's" means Moody's Investors Service, a corporation organized and existing under the laws of the State of Delaware, its successors and assigns, except that if such corporation shall no longer perform the function of a securities rating agency for any reason, the term "Moody's" shall be deemed to refer to any other nationally recognized securities rating agency selected by the Authority.

"Office of the Trustee" means the principal corporate trust office of the Trustee in Los Angeles, California, or such other office as may be specified to the Authority and the City by the Trustee in writing, except that with respect to presentation of Bonds for payment or for registration of transfer and exchange such term shall mean the office or the agency of the Trustee at which, at any particular time, its corporate trust agency shall be conducted as specified to the Authority and the City by the Trustee in writing.

"**Opinion of Counsel**" means a written opinion of counsel of recognized national standing in the field of law relating to municipal bonds, appointed and paid by the Authority or the City and which written opinion is satisfactory to the Trustee.

"Outstanding," when used as of any particular time with reference to Bonds, means (subject to the provisions of Section 11.09 hereof) all Bonds theretofore, or thereupon being, authenticated and delivered by the Trustee under this Indenture except:

(a) Bonds theretofore canceled by the Trustee or surrendered to the Trustee for cancellation;

- (b) Bonds with respect to which all liability of the Authority shall have been discharged in accordance with Section 10.01 hereof; and
- (c) Bonds for the transfer or exchange of or in lieu of or in substitution for which other Bonds shall have been authenticated and delivered by the Trustee pursuant to this Indenture.

"Owner" means, with respect to a Bond, the Person in whose name such Bond is registered on the Registration Books.

"Participant" means any entity which is recognized as a participant by DTC in the bookentry system of maintaining records with respect to Book-Entry Bonds.

"Participating Underwriter" means, with respect to the Series BG Bonds, B.C. Ziegler and Company, and with respect to each issue of Additional Bonds, the institution or institutions, if any, with whom the Authority enters into a purchase contract for the sale of such issue.

"**Permitted Investments**" means any of the following to the extent then permitted by the general laws of the State of California:

(1) (a) Direct obligations (other than an obligation subject to—Variation in principal repayment) of the United States of America ("United States Treasury Obligations"), (b) obligations fully and unconditionally guaranteed as to timely payment of principal and interest by the United States of America, (c) obligations fully and unconditionally guaranteed as to timely payment of principal and interest by any agency or instrumentality of the United States of America when such obligations are backed by the full faith and credit of the United States of America, or (d) evidences of ownership of proportionate interests in future interest and principal payments on obligations described above held by a bank or trust company as custodian, under which the owner of the investment is the real party in interest and has the right to proceed directly and individually against the obligor and the underlying government obligations are not available to any person claiming through the custodian or to whom the custodian may be obligated (collectively "United States Obligations"). These include, but are not necessarily limited to:

-U.S. Treasury obligations

All direct or fully guaranteed obligations

-General Services Administration

Participation certificates

-U.S. Maritime Administration

Guaranteed Title XI financing

-Small Business Administration

Guaranteed participation certificates

Guaranteed pool certificates

-Government National Mortgage Association (GNMA)

GNMA-guaranteed mortgage-backed securities

GNMA-guaranteed participation certificates

-U.S. Department of Housing & Urban Development

Local authority bonds

-Washington Metropolitan Area Transit Authority

Guaranteed transit bonds

- (2) Federal Housing Administration debentures.
- (3) The listed obligations of government-sponsored agencies which are not backed by the full faith and credit of the United States of America:
  - -Federal Home Loan Mortgage Corporation (FHLMC)

Participation certificates (excluded are stripped mortgage securities which are purchased at prices exceeding their principal amounts)

Senior debt obligations

-Farm Credit Banks (formerly: Federal Land Banks, Federal intermediate

Credit Banks and Banks for Cooperatives)

Consolidated systemwide bonds and notes

-Federal Home Loan Banks (FHL Banks)

Consolidated debt obligations

-Federal National Mortgage Association (FNMA)

Senior debt obligations

Mortgage-backed securities (excluded are stripped mortgages securities which are purchased at prices exceeding their principal amounts)

-Financing Corporation (FICO)

Debt obligations

-Resolution Funding Corporation (REFCORP)

Debt obligations

- (4) Unsecured certificates of deposit, time deposits, and bankers' acceptances (having maturities of not more than 180 days) of any bank, including the Trustee and its affiliates, the short-term obligations of which are rated "A-1+" or better by S&P and "P-1" or better by Moody's.
- (5) Deposits the aggregate amount of which are fully insured by the Federal Deposit Insurance Corporation (FDIC), in banks, including the Trustee and its affiliates, which have capital and surplus of at least \$5 million.
- (6) Commercial paper (having original maturities of not more than 270 days) rated "A-1+" by S&P and "Prime-1" by Moody's.
- (7) Money market funds rated "AAm" or "AAm-G" or better by S&P and "Aa2" or better by Moody's, including funds for which the Trustee, its parent holding company, if any, or any affiliates or subsidiaries of the Trustee provide investment advisory or other management services.

## (8) Repurchase agreements:

(a) With any domestic bank the long term debt of which is rated "AA" or better by S&P and "Aa" by Moody's (so long as an opinion is rendered that the repurchase agreement is a "repurchase agreement" as defined in the Financial Institutions Reform, Recovery and Enforcement Act of 1989 ("FIRREA") and that such bank is subject to FIRREA), or any foreign bank rated at least "AA" by S&P and "Aaa" by Moody's or "AA" by S & P and at least "Aa2" by Moody's; provided the term of such repurchase agreement is for one year or less.

- (b) With (i) any broker-dealer with "retail customers" which has, or the parent company of which has, long-term debt rated at least "AA" by S&P and "Aa2" by Moody's, which broker-dealer falls under the jurisdiction of the Securities Investors Protection Corp. (SIPC); provided that:
  - A. The market value of the collateral is maintained for United States Treasury Obligations, at the levels shown below under "Collateral Levels for United States Treasury Obligations";
  - B. Failure to maintain the requisite collateral percentage will require the City or the Trustee to liquidate the collateral;
  - C. The Trustee, the City or a third party acting solely as agent therefor (the "Holder of the Collateral") has possession of the collateral or the collateral has been transferred to the Holder of the Collateral in accordance with applicable state and federal laws (other than by means of entries on the transferor's books);
  - D. The repurchase agreement states, and an opinion of counsel is rendered to the effect, that the Trustee has a perfected first priority security interest in the collateral, any substituted collateral and all proceeds thereof (in the case of bearer securities, this means the Holder of the Collateral is in possession);
  - E. The transferor represents that the collateral is free and clear of any third-party liens or claims;
  - F. An opinion is rendered that the repurchase agreement is a "repurchase agreement" as defined in the United States Bankruptcy Code;
  - G. There is or will be a written agreement governing every repurchase transaction;
  - H. The City represents that it has no knowledge of any fraud involved in the repurchase transaction; and
  - I. The City and the Trustee receive an opinion of counsel (which opinion shall be addressed to the City and the Trustee) that such repurchase agreement is legal, valid and binding and enforceable against the provider in accordance with its terms.

# (9) State Obligations

(a) Direct general obligations of any state of the United States or any subdivision or agency thereof to which is pledged the full faith and credit of a state the unsecured general obligation debt of which is rated "A2" by Moody's and "A" by S&P, or better, or any obligation fully and unconditionally guaranteed by any state, subdivision or agency whose unsecured general obligation debt is so rated.

- (b) Direct, general short-term obligations of any state agency or subdivision described in (a) above and rated "A-1+" by S&P and "Prime-1" by Moody's.
- (c) Special Revenue Bonds (as defined in the United States Bankruptcy Code) of any state, state agency or subdivision described in (a) above and rated "AA" or better by S&P and "Aa2" or better by Moody's.
- (10) Local Agency Investment Fund.
- (11) Investment agreements with a domestic or foreign bank or corporation (other than a life or property casualty insurance company) the long-term debt of which, or, in the case of a guaranteed corporation the long-term debt of the guarantor, or in the case of a monoline financial guaranty insurance company the claims paying ability of the guarantor, is rated at least "AA" by S&P and "Aa2" by Moody's; provided, that prior written notice of an investment in the investment agreement is provided to S&P and, provided, further, by the terms of the investment agreement:
  - (a) interest payments are to be made to the Trustee at times and in amounts as necessary to pay debt service (or, if the investment agreement is for the construction fund, construction draws) on the Bonds;
  - (b) the invested funds are available for withdrawal without penalty or premium, at any time for purposes identified in this Indenture other than acquisition of alternative investment property upon not more than seven days prior notice (which notice may be amended or withdrawn at any time prior to the specified withdrawal date); provided that the Indenture specifically requires the Trustee or the City to give notice in accordance with the terms of the investment agreement so as to receive funds thereunder with no penalty or premium paid;
  - (c) the investment agreement shall state that it is the unconditional and general obligation of, and is not subordinated to any other obligation of, the provider thereof:
  - (d) a guaranteed rate of interest is to be paid on invested funds and all future deposits, if any, required to be made to restore the amount of such funds to the level specified under the Indenture;
  - (e) the Trustee and the City receive the opinion of domestic counsel (which opinion shall be addressed to the City) that such investment agreement is legal, valid and binding and enforceable against the provider in accordance with its terms and of foreign counsel (if applicable);
  - (f) the investment agreement shall provide that if during its term (A) the provider's or the guarantor's rating by either Moody's or S&P is withdrawn or suspended or falls below "AA" or "Aa2", respectively, or, with respect to a foreign bank, below the ratings of such provider at the delivery date of the investment agreement, the provider must, at the direction of the City or the Trustee (acting at the direction of the City) within 10 days of receipt of such direction, either (1) collateralize the investment agreement by delivering or transferring in accordance

with applicable state and federal laws (other than by means of entries on the provider's books) to the Trustee, the City or a Holder of the Collateral, United States Treasury Obligations which are free and clear of any third-party liens or claims at the Collateral Levels set forth below; or (2) repay the principal of and accrued but unpaid interest on the investment (the choice of (1) or (2) above shall be that of the City or Trustee (acting at the direction of the City), as appropriate), and (B) the provider's or the guarantor's rating by either Moody's or S&P is withdrawn or suspended or falls below "A" or "A2," or, with respect to a foreign bank, below "AA" or "Aa2" by S&P or Moody's, as appropriate, the provider must, at the direction of the City or the Trustee (acting at the direction of the City), within 10 days of receipt of such direction, repay the principal of and accrued but unpaid interest on the investment, in either case with no penalty or premium to the City or Trustee;

- (g) the investment agreement shall state, and an opinion of counsel shall be rendered to the effect, that the Trustee has a perfected first priority security interest in the collateral, any substituted collateral and all proceeds thereof (in the case of bearer securities, this means the Trustee is in possession); and
- (h) the investment agreement must provide that if during its term (A) the provider shall default in its payment obligations, the provider's obligation under the investment agreement shall, at the direction of the City or the Trustee, be accelerated and amounts invested and accrued but unpaid interest thereon shall be repaid to the City or Trustee, as appropriate, and (B) the provider shall become insolvent, not pay its debts as they become due, be declared or petition to be declared bankrupt, etc. ("event of insolvency"), the provider's obligations shall automatically be accelerated and amounts invested and accrued but unpaid interest thereon shall be repaid to the City or Trustee, as appropriate.
- (12) Pre-refunded municipal obligations rated "AAA" by S&P and "Aaa" by Moody's meeting the following requirements:
  - (a) the municipal obligations are (i) not subject to redemption prior to maturity or (ii) the trustee for the municipal obligations has been given irrevocable instructions concerning their call and redemption and the issuer of the municipal obligations has covenanted not to redeem such municipal obligations other than as set forth in such instructions;
  - (b) the municipal obligations are secured by cash or United States Treasury Obligations which may be applied only to payment of the principal of, interest and premium on such municipal obligations;
  - (c) the principal of and interest on the United States Treasury Obligations (plus any cash in the escrow) has been verified by the report of independent certified public accountants to be sufficient to pay in full all principal of, interest, and premium, if any, due and to become due on the municipal obligations ("Verification");

- (d) the cash or United States Treasury Obligations serving as security for the municipal obligations are held by an escrow agent or trustee in trust for owners of the municipal obligations; and
- (e) no substitution of a United States Treasury Obligation shall be permitted except with another United States Treasury Obligation and upon delivery of a new Verification; and
- (f) the cash or the United States Treasury Obligations are not available to satisfy any other claims, including those by or against the trustee or escrow agent.

# **Collateral Levels For United States Treasury Obligations**

# Remaining Maturity

Frequency of Valuation	1 year or less	5 years or less	10 years or less	15 years or less	30 years or less
Daily	102	105	106	108	114
Weekly	103	111	112	114	120
Monthly	105	117	120	125	133
Quarterly	107	120	130	133	140

Further Requirements: (a) On each valuation date, the City, or the custodian who shall confirm to the City and the Trustee, shall value the market value (exclusive of accrued interest) of the collateral, which market value will be an amount equal to the requisite collateral percentage times the principal amount of the investment (including unpaid accrued interest thereon) that is being secured, (b) in the event the collateral level is below its collateral percentage on a valuation date, such percentage shall be restored within the following restoration periods: one Business Day for daily valuations, two Business Days for weekly and monthly valuations, and one month for quarterly valuations (the use of different restoration periods affect the requisite collateral percentage), (c) the City or the Trustee (acting at the direction of the City) shall terminate the repurchase agreement or the investment agreement, as the case may be, upon a failure to maintain the requisite collateral percentage after the restoration period and, if not paid by the counterparty in federal funds against transfer of the collateral, liquidate the collateral.

The Trustee shall have no responsibility to monitor the ratings of Permitted Investments after the initial purchase of such Permitted Investments.

"**Person**" means an individual, corporation, firm, association, partnership, trust, or other legal entity or group of entities, including a governmental entity or any agency or political subdivision thereof.

"**Principal Fund**" means the fund by that name established in accordance with Section 5.02 hereof.

"Rebate Fund" means the fund by that name established in accordance with Section 5.06 hereof.

"**Rebate Requirement**" has the meaning ascribed thereto in the Tax Certificate.

- "**Record Date**" means the fifteenth day of the month next preceding an Interest Payment Date, whether or not such day is a Business Day.
- "**Redemption Fund**" means the fund by that name established in accordance with Section 5.02 hereof.
- "**Redemption Price**" means the aggregate amount of principal of and premium, if any, on the Bonds upon the redemption thereof pursuant hereto.
- "**Registration Books**" means the records maintained by the Trustee for the registration of ownership and registration of transfer of the Bonds pursuant to Section 2.05 hereof.
- "Rental Payments" means, collectively, the Base Rental Payments and the Additional Rental Payments.
- "Rental Period" means the period from the Closing Date through [June 30, 2017] and, thereafter, the twelve-month period commencing on July 1 of each year during the term of the Lease Agreement.
- "Representation Letter" means the Letter of Representations from the Authority to DTC, or any successor securities depository for any Series of Book-Entry Bonds, in which the Authority makes certain representations with respect to issues of its securities for deposit by DTC or such successor depository.
- "Reserve Fund" means the fund by that name established and held by the Trustee pursuant to Section 5.05.
- "Reserve Requirement" means, with respect to the Series BG Bonds, an amount equal to \$\_\_\_\_\_ and with respect to any Additional Bonds, such amounts as shall be determined in accordance with the Supplemental Indenture.
- "S&P" means S&P Global Ratings, a Standard & Poor's Financial Services, LLC business, a corporation organized and existing under the laws of the State of New York, its successors and assigns, except that if such entity shall no longer perform the functions of a securities rating agency for any reason, the term "S&P" shall be deemed to refer to any other nationally recognized securities rating agency selected by the Authority.
- "Series" means the Series BG Bonds executed, authenticated and delivered on the Closing Date and identified pursuant to this Indenture and any Additional Bonds issued pursuant to a Supplemental Indenture and identified as a separate Series of Bonds.
- "**Series BG Bonds**" means the Pomona Public Financing Authority 2017 Lease Revenue Refunding Bonds, Series BG (Federally Taxable) issued hereunder.
- "Series BG Reserve Surety Policy" means the municipal bond debt service reserve insurance policy issued by the Insurer and deposited in the Reserve Fund to satisfy the Reserve Requirement for the Series BG Bonds.

"Supplemental Indenture" means any supplemental indenture amendatory of or supplemental to this Indenture, but only if and to the extent that such Supplemental Indenture is specifically authorized hereunder.

"**Tax Certificate**" means the tax certificate delivered by the Authority and the City at the time of the issuance and delivery of a Series of Bonds, as the same may be amended or supplemented in accordance with its terms.

"Trustee" means Zions Bank, a division of ZB, N.A., a national banking association duly organized and existing under and by virtue of the laws of the United States, or any successor thereto as Trustee hereunder, appointed as provided herein.

"Written Certificate of the Authority" and "Written Request of the Authority" mean, respectively, a written certificate or written request signed in the name of the Authority by an Authorized Authority Representative. Any such certificate or request may, but need not, be combined in a single instrument with any other instrument, opinion or representation, and the two or more so combined shall be read and construed as a single instrument

"Written Certificate of the City" and "Written Request of the City" mean, respectively, a written certificate or written request signed in the name of the City by an Authorized City Representative. Any such certificate or request may, but need, not, be combined in a single instrument with any other instrument, opinion or representation, and the two or more so combined shall be read and construed as a single instrument.

1.02 Equal Security. In consideration of the acceptance of the Bonds by the Owners thereof, this Indenture shall be deemed to be and shall constitute a contract among the Authority, the City, the Trustee and the Owners from time to time of all Bonds authorized, executed, issued and delivered hereunder and then Outstanding to secure the full and final payment of the principal of, premium, if any, and interest on all Bonds which may from time to time be authorized, executed, issued and delivered hereunder, subject to the agreements, conditions, covenants and provisions contained herein; and all agreements and covenants set forth herein to be performed by or on behalf of the Authority or the City shall be for the equal and proportionate benefit, protection and security of all Owners of the Bonds without distinction, preference or priority as to security or otherwise of any Bonds over any other Bonds by reason of the number or date thereof or the time of authorization, sale, execution, issuance or delivery thereof or for any cause whatsoever, except as expressly provided herein or therein.

## **ARTICLE II**

# THE BONDS

**2.01 Authorization of Bonds**. The Authority hereby authorizes the issuance of the Bonds under and subject to the terms of this Indenture and applicable laws of the State of California for the purpose of refunding the Refunded Bonds. The Bonds may consist of one or more Series of Bonds of varying denominations, dates, maturities, interest rates and other provisions, subject to the provisions and conditions contained herein.

## 2.02 Terms of Series BG Bonds.

- (a) The Series BG Bonds shall be designated the "Pomona Public Financing Authority 2017 Lease Revenue Refunding Bonds, Series BG (Federally Taxable)." Each Series of Additional Bonds shall bear such additional designation as may be necessary or appropriate to distinguish such Series from every other Series, of Bonds.
- (b) The Series BG Bonds shall be issued in fully registered form without coupons in Authorized Denominations, so long as no Series BG Bond shall have more than one maturity date. The Series BG Bonds shall be dated as of the Closing Date, shall be issued in the aggregate principal amount of \$\_\_\_\_\_\_, shall mature on July 1 of each year and shall bear interest (calculated on the basis of a 360-day year comprised of twelve 30-day months) at the rates per annum as follows:

Maturity Date	Principal				
(July 1)	Amount	Interest Rate			
	\$	%			

Date next preceding the date of authentication thereof unless (i) a Series BG Bond is authenticated on or before an Interest Payment Date and after the close of business on the preceding Record Date, in which event it shall bear interest from such Interest Payment Date, (ii) a Series BG Bond is authenticated on or before the first Record Date, in which event interest thereon shall be payable from the dated date thereof, or (iii) interest on any Series BG Bond is in default as of the date of authentication thereof, in which event interest thereon shall be payable from the date to which interest has been paid in full, payable on each Interest Payment Date. Interest shall be paid in lawful money of the United States on each Interest Payment Date to the Persons in whose names the ownership of the Series BG Bonds is registered on the Registration Books at the close of business on

the immediately preceding Record Date, except as provided below. Interest shall be paid by check of the Trustee mailed by first class mail, postage prepaid, on each Interest Payment Date to the Series BG Bond Owners at their respective addresses shown on the Registration Books as of the close of business on the preceding Record Date.

- (d) The principal and premium, if any, of the Series BG Bonds shall be payable in lawful money of the United States of America upon presentation and surrender thereof upon maturity or earlier redemption at the Office of the Trustee. So long as the Series BG Bonds are registered to Cede & Co., payment of principal and any premiums shall be made without presentment.
- (e) The Series BG Bonds shall be subject to redemption as provided in Article IV.
- **2.03 Form of Series BG Bonds**. The Series BG Bonds shall be in substantially the form set forth in Exhibit A hereto, with appropriate or necessary insertions, omissions and variations as permitted or required hereby.
- 2.04 Transfer and Exchange of Bonds. Any Bond may, in accordance with its terms, be transferred upon the Registration Books by the Person in whose name it is registered, in person or by his duly authorized attorney, upon surrender of such Bond for cancellation, accompanied by delivery of a written instrument of transfer, duly executed in a form acceptable to the Trustee. Whenever any Bond or Bonds shall be surrendered for transfer, the Authority shall execute and the Trustee shall authenticate and shall deliver a new Bond or Bonds of the same Series in a like aggregate principal amount, in any Authorized Denomination. The Trustee shall require the Bond Owner requesting such transfer to pay any tax or other governmental charge required to be paid with respect to such transfer.

The Bonds may be exchanged at the Office of the Trustee for a like aggregate principal amount of Bonds of the same Series of other authorized denominations. The Trustee shall require the payment by the Bond Owner requesting such exchange of any tax or other governmental charge required to be paid with respect to such exchange.

The Trustee shall not be obligated to make any transfer or exchange of Bonds of a Series pursuant to this Section during the period established by the Trustee for the selection of Bonds of such Series for redemption, or with respect to any Bonds of such Series selected for redemption.

- **2.05 Registration Books**. The Trustee will keep or cause to be kept, at the Office of the Trustee, sufficient records for the registration and transfer of ownership of the Bonds, which shall be open to inspection during regular business hours and upon reasonable notice by the City; and, upon presentation for such purpose, the Trustee shall, under such reasonable regulations as it may prescribe, register or transfer or cause to be registered or transferred, on such records, the ownership of the Bonds as hereinbefore provided.
- **2.06** Execution of Bonds. The Bonds shall be executed in the name and on behalf of the Authority with the facsimile signature of an Authorized Officer of the Authority attested by the manual or facsimile signature of the Secretary of the Authority. The Bonds shall then be delivered to the Trustee for authentication by it. In case any of such officers of the Authority who shall have signed or attested any of the Bonds shall cease to be such officers of the Authority before the Bonds

so signed or attested shall have been authenticated or delivered by the Trustee, or issued by the Authority, such Bonds may nevertheless be authenticated, delivered and issued and, upon such authentication, delivery and issue, shall be as binding upon the Authority as though those who signed and attested the same had continued to be such officers of the Authority, and also any Bonds may be signed and attested on behalf of the Authority by such Persons as at the actual date of execution of such Bonds shall be the proper officers of the Authority although at the nominal date of such Bonds any such Person shall not have been such officer of the Authority.

- **2.07 Authentication of Bonds**. Only such of the Bonds as shall bear thereon a certificate of authentication substantially in the form as that set forth in Exhibit A hereto, manually executed by the Trustee, shall be valid or obligatory for any purpose or entitled to the benefits of this Indenture, and such certificate of or on behalf of the Trustee shall be conclusive evidence that the Bonds so authenticated have been duly executed, authenticated and delivered hereunder and are entitled to the benefits of this Indenture.
- **2.08 Temporary Bonds**. The Bonds of a Series may be issued in temporary form exchangeable for definitive Bonds of such Series when ready for delivery. Any temporary Bonds may be printed, lithographed or typewritten, shall be of such authorized denominations as may be determined by the Authority, shall be in fully registered form without coupons and may contain such reference to any of the provisions of this Indenture as may be appropriate. Every temporary Bond shall be executed by the Authority and authenticated by the Trustee upon the same conditions and in substantially the same manner as the definitive Bonds. If the Authority issues temporary Bonds of a Series it will execute and deliver definitive Bonds of such Series as promptly thereafter as practicable, and thereupon the temporary Bonds of such Series, may be surrendered, for cancellation, at the Office of the Trustee and the Trustee shall authenticate and deliver in exchange for such temporary Bonds an equal aggregate principal amount of definitive Bonds of such Series in Authorized Denominations. Until so exchanged, the temporary Bonds of such Series authenticated and delivered hereunder.
- 2.09 Bonds Mutilated, Lost, Destroyed or Stolen. If any Bond shall become mutilated, the Authority, at the expense of the Owner of said Bond, shall execute, and the Trustee shall thereupon authenticate and deliver, a new Bond of like tenor and Series in exchange and substitution for the Bond so mutilated, but only upon surrender to the Trustee of the Bond so mutilated. Every mutilated Bond so surrendered to the Trustee shall be canceled by it and delivered to, or in accordance with the order of, the Authority. If any Bond shall be lost, destroyed or stolen, evidence of such loss, destruction or theft may be submitted to the Trustee and, if such evidence and indemnity satisfactory to the Trustee shall be given, the Authority, at the expense of the Owner, shall execute, and the Trustee shall thereupon authenticate and deliver, a new Bond of like tenor and Series in lieu of and in replacement for the Bond so lost, destroyed or stolen (or if any such Bond shall have matured or shall have been selected for redemption, instead of issuing a replacement Bond, the Trustee may pay the same without surrender thereof). The Authority may require payment by the Owner of a sum not exceeding the actual cost of preparing each replacement Bond issued under this Section and of the expenses which may be incurred by the Authority and the Trustee. Any Bond of a Series issued under the provisions of this Section in lieu of any Bond of such Series alleged to be lost, destroyed or stolen shall constitute an original additional contractual obligation on the part of the Authority whether or not the Bond so alleged to be lost, destroyed or stolen be at any time enforceable by anyone, and shall be entitled to the benefits of this Indenture with all other Bonds of such Series secured by this Indenture.

## **2.10** Book-Entry Bonds.

(a) Prior to the issuance of a Series of Bonds, the Authority may provide that such Series of Bonds shall initially be issued as Book-Entry Bonds and, in such event, the Bonds of such Series for each maturity shall be in the form of a separate single fully registered Bond (which may be typewritten). The Series BG Bonds shall initially be issued as Book-Entry Bonds.

Except as provided in subsection (c) of this Section, the registered Owner of all of the Book-Entry Bonds shall be Cede & Co., as nominee of DTC. Notwithstanding anything to the contrary contained in this Indenture, payment of interest with respect to any Book-Entry Bond registered as of each Record Date in the name of Cede & Co. shall be made by wire transfer of sameday funds to the account of Cede & Co. on the Interest Payment Date at the address indicated on the Record Date for Cede & Co. in the Registration Books or as otherwise provided in the Representation Letter.

- (b) The Trustee and the Authority may treat DTC (or its nominee) as the sole and exclusive Owner of Book-Entry Bonds registered in its name for the purposes of payment of the principal, premium, if any, or interest with respect to Book-Entry Bonds, selecting Book-Entry Bonds or portions thereof to be redeemed, giving any notice permitted or required to be given to Owners of Book-Entry Bonds under this Indenture, registering the transfer of Book-Entry Bonds, obtaining any consent or other action to be taken by Owners of Book-Entry Bonds and for all other purposes whatsoever, and neither the Trustee nor the Authority shall be affected by any notice to the contrary. Neither the Trustee nor the Authority shall have any responsibility or obligation to any Participant, any person claiming a beneficial ownership interest in Book-Entry Bonds under or through DTC or any Participant, or any other person which is not shown on the Registration Books as being an Owner, with respect to the accuracy of any records maintained by DTC or any Participant, the payment by DTC or any Participant of any amount in respect of the principal, premium, if any, or interest with respect to Book-Entry Bonds, any notice which is permitted or required to be given to Owners of Book-Entry Bonds under this Indenture, the selection by DTC or any Participant of any person to receive payment in the event of a partial redemption of Book-Entry Bonds, or any consent given or-other action taken by DTC as Owner of Book-Entry Bonds. The Trustee shall pay all principal, premium, if any and interest with respect to Book-Entry Bonds, only to DTC, and all such payments shall be valid and effective to fully satisfy and discharge the Authority's obligations with respect to the principal, premium, if any, and interest with respect to the Book-Entry Bonds to the extent of the sum or sums so paid. Except under the conditions of subsection (c) of this Section, no person other than DTC shall receive an executed Book-Entry Bond for each separate stated maturity. Upon delivery by DTC to the Trustee of written notice to the effect that DTC has determined to substitute a new nominee in place of Cede & Co., and subject to the provisions herein with respect to record dates, the term "Cede & Co." in this Indenture shall refer to such new nominee of DTC.
- (c) In the event (i) DTC, including any successor as securities depository for a Series of Bonds, determines not to continue to act as securities depository for such Series of Bonds, or (ii) the Authority determines that the incumbent securities depository shall no longer so act, and delivers a written certificate to the Trustee to that effect, then the Authority will discontinue the book-entry system with the incumbent securities depository for such Series of Bonds. If the Authority determines to replace the incumbent securities depository for such Series of Bonds with another qualified securities depository, the Authority shall prepare or direct the preparation of a new single, separate fully registered Bond of such Series for the aggregate outstanding principal amount

of Bonds of such Series of each maturity, registered in the name of such successor or substitute qualified securities depository, or its nominee, or make such other arrangement acceptable to the Authority, the Trustee and the successor securities depository for the Bonds of such Series as are not inconsistent with the terms of this Indenture. If the Authority fails to identify another qualified successor securities depository for such Series of Bonds to replace the incumbent securities depository, then the Bonds of such Series shall no longer be restricted to being registered in the Registration Books in the name of the incumbent securities depository or its nominee, but shall be registered in whatever name or names the incumbent securities depository for such Series of Bonds, or its nominee, shall designate. In such event the Authority shall execute, and deliver to the Trustee, a sufficient quantity of Bonds of such Series to carry out the transfers and exchanges provided in Sections 2.04, 2.08 and 2.09 hereof. All such Bonds of such Series shall be in fully registered form in Authorized Denominations.

- (d) Notwithstanding any other provision of this Indenture to the contrary, so long as any Book-Entry Bond is registered in the name of DTC, or its nominee, all payments with respect to the principal, premium, if any, and interest with respect to such Book-Entry Bond and all notices with respect to such Book-Entry Bond shall be made and given, respectively, as provided in the Representation Letter.
- (e) In connection with any notice or other communication to be provided to Owners of Book-Entry Bonds pursuant to this Indenture by the Authority, the City or the Trustee with respect to any consent or other action to be taken by Owners, the Authority, the City or the Trustee, as the case may be, shall establish a record date for such consent or other action and give DTC notice of such record date not less than 15 calendar days in advance of such record date to the extent possible.

#### **ARTICLE III**

# ISSUANCE OF BONDS; APPLICATION OF PROCEEDS

**3.01 Issuance of Series BG Bonds**. The Authority may, at any time, execute the Series BG Bonds for issuance hereunder and deliver the same to the Trustee. The Trustee shall authenticate the Series BG Bonds and deliver the Series BG Bonds to the original purchaser thereof upon receipt of a Written Request of the Authority and upon receipt of the purchase price therefor.

3.02	Applica	tuon	of Proceeds	oi tn	e Serie	SBG B	onas.	On the	Closing	Date, the
proceeds of th	ie sale c	of the	e Series BG E	onds 1	received	l by the	Trustee	e, \$	(	being the
principal amou	unt of S	\$	, less	net o	original	issue o	discount	of \$		, less an
underwriter's	discount	of	\$,	and	less \$_		paid	l by the	purchas	er of the
Series BG Bonas follows:	ds on be	half (	of the Authorit	y direc	tly to th	e Insure	er), shall	be depos	ited by th	ie Trustee
Issuance Fund.	` /	The	Trustee shall	depos	it the a	amount	of \$		_ in the	Costs of
Escrow Fund.	(b)	The	Trustee shall	transfe	er the a	mount	of \$	i	nto the	Series AR

Escrow Fund.	(c)	The	Trustee	shall	transfer the	amount	of \$	into	the S	Series AU
Escrow Fund.	(d)	The	Trustee	shall	transfer the	amount	of \$	into	the S	Series AV
3.03 designated the of Issuance I deposited in Additional Bo which such A	e "Costs Fund the the Cos ands rec	of Iss e amo at of Is quired	uance Fu unt spec ssuance to be de	ind." ified Fund eposit	in Section 3 the portion,	ing Date, 3.02 hered if any, o	there shal of. There of the pro	l be deposite e shall be a oceeds of th	ed in dditi e sa	the Costs onally be le of any
The matime to time to time to time stating (a) the which the oblissuance Fund Costs of Issuathereunder. Costs of Issuathereunder Costs of Issuathe	to pay to Person ligation l, and (ende Fur Dn [drawn the the Pur Dn first to pay the pay t	to who was in the	sts of Issom paym neurred, such amo each case 1, 201 om by th	suance in the control of the control	s to be made, hat such pays have not bee other with a st ll amounts, if	(b) the arment is a n the subject any, remarks	a Written mount to b proper ch ect of a pr or invoice aining in t	Request of be paid, (c) tharge against rior disburse for each amothe Costs of	the put the ment ount Issua	Authority urpose for Costs of t from the requested ance Fund
3.04	[Rese	rved.]								
3.05 issue one or a Base Rental hereunder, but to the issuance	more Se Paymen t only si	eries of ts as abject	f Addition of Addi	onal l l her llowi	ein on a par	dition to trity with	the Series all other	BG Bonds) Bonds there	pay: etofo	able from ore issued
pursuant here the following:					nch Additiona vided for by a					
Bonds;			(A)	Th	e application	of the pro	oceeds of t	the sale of su	ich A	Additional
Additional Bo	onds and	the de	(B) enominat		e principal r denomination		_		ıch	Series of
dates and the			(C)	Th	e date, the i	naturity (	late or da	ates the int	erest	payment

shall be payable thereafter semiannually on July 1 and January 1, (iii) all Additional Bonds of a Series of like maturity shall be identical in all respects, except as to number or denomination, and (iv) serial maturities of serial Bonds or mandatory sinking fund redemptions for term Bonds, or any combination thereof, shall be established to provide for the redemption or payment of such Additional Bonds on or before their respective maturity dates;

- (D) The redemption premiums and terms, if any, for such Additional Bonds;
  - (E) The form of such Additional Bonds;
- (F) The amount to be deposited in an account maintained in the Reserve Fund for the Additional Bonds so that following such deposit there shall be on deposit in such account of the Reserve Fund an amount at least equal to the Reserve Requirement for the Additional Bonds; and
- (G) Such other provisions that are appropriate or necessary and are not inconsistent with the provisions hereof;
- (b) The Authority shall be in compliance with all agreements, conditions, covenants and terms contained herein, in the Lease Agreement and in the Ground Lease required to be observed or performed by it;
- (c) The City shall be in compliance with all agreements, conditions, covenants and terms contained herein, in the Lease Agreement and in the Ground Lease required to be observed or performed by it; and
- (d) The Ground Lease shall have been amended, to the extent necessary, and the Lease Agreement shall have been amended so as to increase the Base Rental Payments payable by the City thereunder by an aggregate amount equal to the principal of and interest on such Additional Bonds, payable at such times and in such manner as may be necessary to provide for the payment of the principal of and interest on such Additional Bonds; provided, however, that no such amendment shall be made such that the sum of Base Rental Payments, including any increase in the Base Rental Payments as a result of such amendment, plus Additional Rental Payments, in any Rental Period shall be in excess of the annual fair rental value of the Property after taking into account the use of the proceeds of any Additional Bonds issued in connection therewith (evidence of the satisfaction of such condition shall be made by a Written Certificate of the City).

Notwithstanding satisfaction of the other conditions to the issuance of Additional Bonds set forth in this Section 3.05, no such issuance may occur if (1) an event of default (within the meaning of Article VI of the Lease Agreement) (or any event which, once all notice or grace periods have passed, would constitute such event of default) exists unless such event of default shall be cured upon such execution or issuance or (2) the Reserve Fund is not fully funded at the Reserve Requirement for all Outstanding Series of Bonds.

**3.06 Procedure for the Issuance of Additional Bonds**. At any time after the sale of any Additional Bonds in accordance with the Act, such Additional Bonds shall be executed by the Authority for issuance hereunder and shall be delivered to the Trustee and thereupon shall be authenticated and delivered by the Trustee, but only upon receipt by the Trustee of the following:

- (a) Certified copies of the Supplemental Indenture authorizing the issuance of such Additional Bonds, the amendment to the Lease Agreement requited by Section 3.06 hereof and the amendment to the Ground Lease, if any, required by Section 3.06 hereof, together with satisfactory evidence that such amendment to the Lease Agreement and such amendment to the Ground Lease, if any, have been duly recorded;
- (b) A Written Request of the Authority as to the delivery of such Additional Bonds;
- An opinion of Bond Counsel substantially to the effect that (i) the Indenture (c) (including all Supplemental Indentures), the Lease Agreement (including the amendment thereto required by Section 3.06 hereof) and the Ground Lease (including any amendment thereto required by Section 3.06 hereof) have been duly authorized, executed and delivered by, and constitute the valid and binding obligations of, the Authority and the City, enforceable in accordance with their terms (except as enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws affecting creditors rights and by the application of equitable principles and by the exercise of judicial discretion in appropriate cases and subject to the limitations on legal remedies against political subdivisions in the State of California), and (ii) such Additional Bonds constitute valid and binding special obligations of the Authority payable solely from Base Rental Payments as provided herein and are enforceable in accordance with their terms (except as enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws affecting creditors rights and by the application of equitable principles and by the exercise of judicial discretion in appropriate cases and subject to the limitations on legal remedies against political subdivisions in the State of California);
- (d) a Written Certificate of the Authority that the requirements of Section 3.06 hereof have been met;
- (e) a Written Certificate of the City that the requirements of Section 3.06 hereof and Sections 5.01 and 5.02 of the Lease Agreement have been met, and a Written Certificate of the City as to the fair rental value of the Property, after giving effect to the execution and delivery of the Additional Bonds, and to the use of proceeds received therefrom; and
- (f) Such further documents as are required by the provisions hereof or by the provisions of the Supplemental Indenture authorizing the issuance of such Additional Bonds.
- **3.07 Additional Bonds**. So long as any of the Bonds remain Outstanding, the Authority shall not issue any Additional Bonds or obligations payable from the Base Rental Payments, except pursuant to Sections 3.06 and 3.07 hereof.

## **ARTICLE IV**

#### REDEMPTION OF BONDS

# 4.01 Redemption of Series BG Bonds.

(a) <u>Extraordinary Redemption</u>. The Series BG Bonds shall be subject to redemption, in whole or in part, on any date, in Authorized Denominations, from and to the extent of any Net Insurance Proceeds received with respect to all or a portion of the Property, deposited by the

Trustee in the Redemption Fund pursuant to Sections 5.03 and 5.04 hereof, at a Redemption Price equal to the principal amount of the Series BG Bonds to be redeemed, plus accrued interest thereon to the date of redemption, without premium.

- (b) Optional Redemption. The Series BG Bonds maturing on or after July 1, 20\_\_, shall be subject to optional redemption, in whole or in part, on any date on or after July 1, 20\_\_, in Authorized Denominations, from and to the extent of prepaid Base Rental Payments paid pursuant to subsection (a) of Section 7.02 of the Lease Agreement, at a Redemption Price equal to the principal amount of the Series BG Bonds to be redeemed, plus accrued interest thereon to the date of redemption, without premium.
- (c) <u>Sinking Fund Redemption</u>. The Series BG Bonds maturing on July 1, 20\_ are subject to mandatory sinking fund redemption in part (by lot) on each July 1 on and after July 1, 20\_, in integral multiples of \$5,000 at a Redemption Price of the principal amount thereof plus accrued interest to the date fixed for redemption, without premium, in accordance with the following schedule:

Sinking Fund Redemption Date (July 1)

Principal Amount To Be Redeemed

(maturity)

The Series BG Bonds maturing on July 1, 20\_\_ are subject to mandatory sinking fund redemption in part (by lot) on each July 1 on and after July 1, 20\_\_, in integral multiples of \$5,000 at a Redemption Price of the principal amount thereof plus accrued interest to the date fixed for redemption, without premium, in accordance with the following schedule:

Sinking Fund Redemption Date (July 1)

Principal Amount
To Be Redeemed

(maturity)

In the event of a redemption pursuant to Section 4.01(a) or (b), the City shall provide the Trustee with a revised mandatory sinking fund schedule giving effect to the redemption so completed.

**4.02 Notice of Redemption**. At least 20 but not more than 60 days prior to the date fixed for redemption the Trustee on behalf and at the expense of the Authority shall mail (by first class mail) notice of any redemption to the respective Owners of any Bonds designated for redemption at their respective addresses appearing on the Registration Books, provided, however, so long as the Bonds are registered in the name of the Nominee, notice shall be given in such manner as complies with the requirements of DTC. Such notice shall state the date of the notice, the redemption date, the

redemption place and the Redemption Price and shall designate the CUSIP numbers, the Bond numbers and the maturity or maturities (except in the event of redemption of all of the Bonds of such maturity or maturities in whole) of the Bonds to be redeemed, and shall require that such Bonds be then surrendered at the Office of the Trustee for redemption at the Redemption Price, giving notice also that further interest on such Bonds will not accrue from and after the date fixed for redemption. Such notice may state that such redemption is conditioned upon sufficient funds being on deposit on the redemption date to redeem the Bonds so called for redemption. Such notice of redemption may also state that no representation is made as to the accuracy or correctness of the CUSIP numbers printed therein or on the Bonds. Neither the failure to receive any notice so mailed, nor any defect in such notice, shall affect the validity of the proceedings for the redemption of the Bonds or the cessation of accrual of interest thereon from and after the date fixed for redemption.

- 4.03 Selection of Bonds for Redemption. Whenever provision is made in this Indenture for the redemption of less than all of the Bonds, the Trustee shall select the Bonds to be redeemed from all Bonds not previously called for redemption (a) with respect to any optional redemption of Bonds of a Series, among maturities of Bonds of such Series as directed in a Written Request of the Authority, (b) with respect to any redemption pursuant to Section 4.01(a) hereof and the corresponding provision of any Supplemental Indenture pursuant to which Additional Bonds are issued, among maturities of all Series of Bonds on a pro rata basis as nearly as practicable, approved in writing by the Insurer (so long as the Insurer is not in default on any obligation under the Insurance Policy), and (c) with respect to any other redemption of Additional Bonds, among maturities as provided in the Supplemental Indenture pursuant to which such Additional Bonds are issued, and by lot among Bonds of the same Series with the same maturity in any manner which the Trustee in its sole discretion shall deem appropriate and fair. For purposes of such selection, all Bonds shall be deemed to be comprised of separate \$5,000 denominations and such separate denominations shall be treated as separate Bonds which may be separately redeemed.
- **4.04 Partial Redemption of Bonds**. Upon surrender of any Bonds redeemed in part only, the Authority shall execute and the Trustee shall authenticate and deliver to the Owner thereof, at the expense of the Authority, a new Bond or Bonds of the same Series in authorized denominations equal in aggregate principal amount representing the unredeemed portion of the Bonds surrendered.
- **4.05 Effect of Notice of Redemption**. Notice having been mailed as aforesaid, and moneys for the Redemption Price, and the interest to the applicable date fixed for redemption, having been set aside in the Redemption Fund, the Bonds shall become due and payable on said date, and, upon presentation and surrender thereof at the Office of the Trustee, said Bonds shall be paid at the Redemption Price thereof, together with interest accrued and unpaid to said date.

If, on said date fixed for redemption, moneys for the Redemption Price of all the Bonds to be redeemed, together with interest to said date, shall be held by the Trustee so as to be available therefor on such date, and, if notice of redemption thereof shall have been mailed as aforesaid and not canceled, then, from and after said date, interest on said Bonds shall cease to accrue and become payable. All moneys held by or on behalf of the Trustee for the redemption of Bonds shall be held in trust for the account of the Owners of the Bonds so to be redeemed without liability to such Owners for interest thereon.

All Bonds paid at maturity or redeemed prior to maturity pursuant to the provisions hereof shall be canceled upon surrender thereof and destroyed.

#### **ARTICLE V**

# SECURITY FOR BONDS; FLOW OF FUNDS; INVESTMENTS

**5.01 Pledge; Special Obligations**. Subject only to the provisions of this Indenture permitting the application thereof for the purposes and on the terms and conditions set forth herein, all of the Base Rental Payments and any other amounts held in the Base Rental Payment Fund, the Interest Fund, the Principal Fund and the Redemption Fund are hereby pledged to secure the payment of the principal of, premium, if any, and interest on the Bonds in accordance with their terms, the provisions of this Indenture and the Act. Said pledge shall constitute a first lien on such assets.

All obligations of the Authority under this Indenture shall be special obligations of the Authority, payable solely from Rental Payments and the other assets pledged therefor hereunder; provided, however, that all obligations of the Authority under the Bonds shall be special obligations of the Authority, payable solely from Base Rental Payments and the other assets pledged therefor hereunder. Neither the faith and credit nor the taxing power of the Authority, the City or the State of California, or any political subdivision thereof, is pledged to the payment of the Bonds.

## 5.02 Flow of Funds.

(a) The Trustee shall establish and maintain separate funds designated the "Base Rental Payment Fund," the "Interest Fund," the "Principal Fund," the "Reserve Fund" and the "Redemption Fund." Within the Base Rental Payment Fund, the Trustee shall establish the "Prepaid Base Rental Account."

All Base Rental Payments shall be paid directly by the City to the Trustee, and if received by the Authority at any time shall be transferred by the Authority with the Trustee within one Business Day after the receipt thereof. All Base Rental Payments received by the Trustee shall be deposited by the Trustee in the Base Rental Payment Fund.

- (b) The Trustee shall transfer the amounts on deposit in the Base Rental Payment Fund, at the times and in the manner hereinafter provided, to the following respective funds:
- (i) <u>Interest Fund</u>. On the Business Day immediately preceding each Interest Payment Date, the Trustee shall transfer from the Base Rental Payment Fund to the Interest Fund the amount, if any, necessary to cause the amount on deposit in the Interest Fund to be equal to the interest due on the Bonds on such Interest Payment Date.
- (ii) <u>Principal Fund</u>. On the Business Day immediately preceding each July 1, commencing [July 1, 2017], the Trustee shall transfer from the Base Rental Payment Fund to the Principal Fund the amount, if any, necessary to cause the amount on deposit in the Principal Fund to be equal to the principal amount of the Bonds due on such July 1, either as a result of the maturity thereof or mandatory sinking fund redemption payments required to be made with respect thereto. Moneys in the Principal Fund shall be used by the Trustee for the purpose of paying the principal of the Bonds when due and payable at their maturity dates or upon earlier mandatory sinking fund redemption.
- (iii) <u>Redemption Fund</u>. The Trustee, on the redemption date specified in the Written Request of the City filed with the Trustee at the time that any prepaid Base Rental

Payment is paid to the Trustee pursuant to the Lease Agreement, shall deposit in the Redemption Fund that amount of moneys representing the portion of the Base Rental Payments designated as prepaid Base Rental Payments. Additionally, the Trustee shall deposit in the Redemption Fund any amounts required to be deposited therein pursuant to Section 5.03 or Section 5.04 hereof. Moneys in the Redemption Fund shall be used by the Trustee for the purpose of paying the principal of and interest and premium, if any, on Series BG Bonds redeemed pursuant to the provisions of subsections (a) and (b) of Section 4.01 hereof and Additional Bonds redeemed pursuant to the corresponding provisions of the Supplemental Indenture pursuant to which such Additional Bonds are issued.

- (c) Upon receipt of a Written Certificate of the City pursuant to Section 3.06 of the Lease Agreement that the City has commenced repairs of the Property, the Trustee will transfer moneys from the Prepaid Base Rental Account at the times and in the manner required by subsection (b) above if and to the extent there are insufficient funds in the Base Rental Payment Fund to make such transfers. Upon receipt of a Written Certificate of the City pursuant to Section 3.06 of the Lease Agreement that the City has completed repairs of the Property and will recommence to make Base Rental Payments, or that the City has determined not to make such repairs or such repairs do not require the City to vacate the Property, and made the certifications required pursuant to the Lease Agreement, the Trustee will transfer any funds remaining on deposit in the Prepaid Base Rental Account to the City for any lawful use.
- (d) <u>Reserve Fund</u>. All amounts on deposit in the Base Rental Payment Fund on July 2, to the extent not required to pay the principal of any Outstanding Bonds then having come due and payable, shall be transferred to the Reserve Fund, to the extent necessary to satisfy the Reserve Requirement or to pay Policy Costs to the Insurer (as such terms are defined herein), to the extent such amounts are due and payable.
- (e) <u>Insurer Reimbursement Amounts</u>. The Authority shall from the remaining moneys in the Base Rental Payment Fund pay Insurer Reimbursement Amounts to the Insurer (as such terms are defined herein) not payable from the Interest Fund or Principal Fund, to the extent such amounts are due and payable.
- (f) <u>Surplus</u>. Any amounts remaining in the Base Rental Payment Fund shall be withdrawn therefrom and transferred to the City to be used for any lawful purposes of the City.
- **5.03 Application of Net Insurance Proceeds**. If the Property or any portion thereof shall be damaged or destroyed, subject to the further requirements of this Section, the City shall, as expeditiously as possible, continuously and diligently prosecute or cause to be prosecuted the repair or replacement thereof, unless the City elects not to repair or replace the Property or the affected portion thereof in accordance with the provisions hereof.

The Net Insurance Proceeds (other than Net Insurance Proceeds of rental interruption insurance), including the proceeds of any self-insurance, received on account of any damage or destruction of the Property or a portion thereof shall as soon as possible be deposited with the Trustee and be held by the Trustee in a special account and made available for and, to the extent necessary, shall be applied to the cost of repair or replacement of the Property or the affected portion thereof upon receipt of a Written Request of the City, together with invoices therefor. Pending such application, such proceeds may be invested by the Trustee as directed by the City in Permitted Investments that mature not later than such times moneys are expected to be needed to pay such costs of repair or replacement.

Notwithstanding the foregoing, the City shall, within 60 days of the occurrence of the event of damage or destruction, notify the Trustee in writing as to whether the City intends to replace or repair the Property or the portions of the Property which were damaged or destroyed. If the City does intend to replace or repair the Property or portions thereof, the City shall deposit with the Trustee the full amount of any insurance deductible to be credited to the special account.

If the damage, destruction or loss was such that there resulted a substantial interference with the City's right to the use or occupancy of the Property and an abatement, of Rental Payments results from such damage or destruction pursuant to Section 3.06 of the Lease Agreement, then the City shall be required either to (a) apply sufficient funds from the insurance proceeds and other legally available funds to the replacement or repair of the Property or the portions thereof which have been damaged to the condition which existed prior to such damage or destruction, or (b) apply sufficient funds from the insurance proceeds and other legally available funds to the redemption, as set forth in subsection (a) of Section 4.01 hereof and the corresponding provisions of any Supplemental Indenture pursuant to which Additional Bonds are issued, in full of all the Outstanding Bonds or all of those Outstanding Bonds which would have been payable from that portion of the Base Rental Payments which are abated as a result of the damage or destruction. Funds to be applied to the redemption of Bonds in accordance with clause (b) above shall be deposited in the Redemption Fund. If the City is not required to replace or repair the Property, or the affected portion thereof, as set forth in clause (a) above or to use such amounts to redeem Bonds as set forth in clause (b) above, then such proceeds shall, if there is first delivered to the Trustee a Written Certificate of the City to the effect that the annual fair rental value of the Property after such damage or destruction, and after any repairs or replacements made as a result of such damage or destruction, is at least equal to 100% of the maximum amount of Base Rental Payments becoming due under the Lease Agreement in the then current Rental Period or any subsequent Rental Period and the fair replacement value of the Property after such damage or destruction is at least equal to the principal amount of the Outstanding Bonds, be paid to the City to be used for any lawful purpose.

Proceeds of rental interruption insurance shall be applied to the payment of Base Rental Payments to the extent of any abatement thereof pursuant to the Lease Agreement, and otherwise as directed in writing by the City.

The proceeds of any award in eminent domain received in respect to the Property shall be deposited by the Trustee in the Redemption Fund and applied to the redemption of Bonds pursuant to subsection (a) of Section 4.01 hereof and the corresponding provisions of any Supplemental Indenture pursuant to which Additional Bonds are issued.

- **5.04 Title Insurance**. Proceeds of any policy of title insurance received by the Trustee in respect of the Property shall be applied and disbursed by the Trustee as follows:
- (a) if the City determines that the title defect giving rise to such proceeds has not substantially interfered with its use and occupancy of the Property and will not result in an abatement of Rental Payments payable by the City under the Lease Agreement, such proceeds shall be remitted to the City and used for any lawful purpose thereof; or
- (b) if the City determines that the title defect giving rise to such proceeds has substantially interfered with its use and occupancy of the Property and will result in an abatement of Rental Payments payable by the City under the Lease Agreement, then the Trustee shall immediately deposit such proceeds in the Redemption Fund and such proceeds shall be applied to the redemption

of Bonds in the manner provided in subsection (a) of Section 4.01 hereof and the corresponding provisions of any Supplemental Indenture pursuant to which Additional Bonds are issued.

- **5.05 Reserve Fund.** The Authority hereby agrees to establish a separate fund titled the "Reserve Fund" and within the Reserve Fund, the "Series BG Bonds Account," which shall be held and maintained by the Trustee. The Trustee shall deposit into the Reserve Fund or accounts therein, any amounts transferred from the Base Rental Fund pursuant to Section 5.02(d) above. The Trustee may establish additional accounts in the Reserve Fund in connection with the issuance of Additional Bonds, if any. All moneys in the Reserve Fund and accounts therein shall be used and withdrawn by the Trustee solely for the purpose of funding the Interest Fund or the Principal Fund, in that order, in the event of any deficiency in either of such accounts on an Interest Payment Date. The Trustee shall deposit the Series BG Reserve Surety Policy into the Series BG Bonds Reserve Account.
- (a) As long as the Series BG Reserve Surety Policy shall be in full force and effect the Authority, the City and the Trustee agree to comply with the following provisions:
- (i) In the event and to the extent that moneys on deposit in the Base Rental Payment Fund, plus all amounts on deposit in and credited to the Reserve Fund in excess of the amount of the Series BG Reserve Surety Policy, are insufficient to pay the amount of principal and interest coming due on the Series BG Bonds, then upon the later of: (i) one (1) Business Day after receipt by the Insurer of a Notice of Nonpayment (as such terms are defined in the Series BG Reserve Surety Policy), duly executed by the Trustee certifying that payment due under this Indenture has not been made to the Trustee; or (ii) the Interest Payment Date, the Insurer will make a deposit of funds in an account with the Trustee or its successor sufficient for the payment to the Trustee of amounts which are then due to the Trustee under this Indenture up to but not in excess of the Policy Limit, as defined in the Series BG Reserve Surety Policy; provided, however, that in the event that the amount on deposit in, or credited to, the Reserve Fund, in addition to the amount available under the Series BG Reserve Surety Policy, includes amounts available under a letter of credit, insurance policy, reserve surety policy or other such funding instrument (the "Additional Funding Instrument"), draws on the Series BG Reserve Surety Policy and the Additional Funding Instrument shall be made on a pro rata basis to fund the insufficiency.
- (ii) The Authority shall repay from Base Rental Payments any draws under the Series BG Reserve Surety Policy and pay all related reasonable expenses incurred by the Insurer. Interest shall accrue and be payable on such draws and expenses from the date of payment by the Insurer at the Late Payment Rate. For purposes of this Section 5.03, "Late Payment Rate" means the lesser of: (i) the greater of: (A) the per annum rate of interest, publicly announced from time to time by JP Morgan Chase Bank at its principal office in the City of New York, as its prime or base lending rate ("Prime Rate") (any change in such Prime Rate to be effective on the date such change is announced by JP Morgan Chase Bank) plus 3%; and (B) the then applicable highest rate of interest on the Series BG Bonds; and (ii) the maximum rate permissible under applicable usury or similar laws limiting interest rates. The Late Payment Rate shall be computed on the basis of the actual number of days elapsed over a year of 360 days. In the event that JP Morgan Chase Bank ceases to announce its Prime Rate publicly, Prime Rate shall be the publicly announced prime or base lending rate of such national bank as the Insurer shall specify.

If the interest provisions of this Section 5.05(a)(ii) shall result in an effective rate of interest which, for any period, exceeds the limit of the usury or any other laws applicable to the indebtedness created herein, then all sums in excess of those lawfully collectible as interest for

the period in question shall, without further agreement or notice between or by any party hereto, be applied as additional interest for any later periods of time when amounts are outstanding hereunder to the extent that interest otherwise due hereunder for such periods plus such additional interest would not exceed the limit of the usury or such other laws, and any excess shall be applied upon principal immediately upon receipt of such moneys by the Insurer, with the same force and effect as if the Authority had specifically designated such extra sums to be so applied and the Insurer had agreed to accept such extra payment(s) as additional interest for such later periods. In no event shall any agreed-to or actual exaction as consideration for the indebtedness created herein exceed the limits imposed or provided by the law applicable to this transaction for the use or detention of money or for forbearance in seeking its collection.

- (iii) Repayment of draws and payment of expenses and accrued interest thereon at the Late Payment Rate (collectively, "Policy Costs") shall commence in the first month following each draw, and each such monthly payment shall be in an amount at least equal to 1/12 of the aggregate of Policy Costs related to such draw.
- (iv) Amounts in respect of Policy Costs paid to the Insurer shall be credited first to interest due, then to the expenses due and then to principal due. As and to the extent that payments are made to the Insurer on account of principal due, the coverage under the Series BG Reserve Surety Policy will be increased by a like amount, subject to the terms of the Series BG Reserve Surety Policy.
- (v) All cash and investments in the Series 2017 Reserve Account shall be transferred to the Base Rental Payment Fund for payment of the principal of and interest on the Series BG Bonds before any drawing may be made on the Series BG Reserve Surety Policy or any other credit facility credited to the Reserve Fund in lieu of cash ("Credit Facility"). Payment of any Policy Costs shall be made prior to replenishment of any such cash amounts. Draws on all Credit Facilities (including the Series BG Reserve Surety Policy) on which there is available coverage shall be made on a pro-rata basis (calculated by reference to the coverage then available thereunder) after applying all available cash and investments in the Series 2017 Reserve Account. Payment of Policy Costs and reimbursement of amounts with respect to other Credit Facilities shall be made on a pro-rata basis prior to replenishment of any cash drawn from the Series 2017 Reserve Account. For the avoidance of doubt, "available coverage" means the coverage then available for disbursement pursuant to the terms of the applicable alternative credit instrument without regard to the legal or financial ability or willingness of the provider of such instrument to honor a claim or draw thereon or the failure of such provider to honor any such claim or draw.
- (vi) If the Authority or the City shall fail to pay any Policy Costs in accordance with the requirements of Section 5.05(a)(ii) hereof, the Insurer shall be entitled to exercise any and all legal and equitable remedies available to it, including those provided under this Indenture other than: (A) acceleration of the maturity of the payments of principal of and interest on the Series BG Bonds; or (B) remedies which would adversely affect Owners of the Series BG Bonds.
- (vii) The Trustee shall ascertain the necessity for a claim upon the Series BG Reserve Surety Policy in accordance with the provisions of Section 5.05(a)(i) hereof and provide notice to the Insurer in accordance with the terms of the Series BG Reserve Surety Policy at least five (5) Business Days prior to an Interest Payment Date. Where deposits are required to be made by the Authority with the Trustee to the Base Rental Payment Fund for the payment of principal of and interest on the Series BG Bonds more often than semi-annually, the Trustee shall be

instructed to give notice to the Insurer of any failure of the Authority to make timely payment in full of such deposits within two Business Days of the date due.

- (viii) The obligation to pay Policy Costs shall be secured by a valid lien on the Base Rental Payments, subject to the priority of payments set forth in this Indenture.
- (ix) So long as the Series BG Reserve Surety Policy is in full force and effect, future deposits of a Credit Facility in the Reserve Fund shall require the prior written consent of the Insurer. Notwithstanding anything in this Indenture to the contrary, amounts on deposit in the Series 2017 Reserve Account shall be applied solely to the payment of principal of and interest on the Series BG Bonds.

Neither this Indenture nor the Lease Agreement shall be discharged until all Policy Costs owing to the Insurer as issuer of the Series BG Reserve Surety Policy have been paid in full.

(b) On the final maturity date for the Bonds, amounts in the Reserve Fund shall be transferred to the Interest Fund and the Principal Fund to make the final payment on the Bonds, and the amount transferred shall be deemed to be a credit against the remaining Base Rental Payments to be made by the City under the Lease Agreement.

## **5.06** [Reserved.]

**5.07 Investment of Moneys**. Except as otherwise provided herein, all moneys in any of the funds or accounts established pursuant to this Indenture and held by the Trustee shall be invested by the Trustee solely in Permitted Investments, as directed in writing by the Authority. Moneys in all funds and accounts held by the Trustee shall be invested in Permitted Investments maturing not later than the date on which it is estimated that such moneys will be required for the purposes specified in this Indenture; provided, however, that such Permitted Investments may be redeemed at par so as to be available on each Interest Payment Date. Absent timely written direction from the Authority, the Trustee shall hold any funds held by it uninvested.

Subject to the provisions of Section 5.07 hereof, all interest, profits and other income received from the investment of moneys in any fund or account established pursuant to this Indenture shall be retained therein.

Permitted Investments acquired as an investment of moneys in any fund established under this Indenture shall be credited to such fund. For the purpose of determining the amount in any fund, all Permitted Investments credited to such fund shall be valued by the Trustee at the fair market value thereof, such valuation to be performed not less frequently than semiannually on or before each March 15 and September 15. In determining fair market value, the Trustee may use and rely conclusively on any generally recognized securities pricing service available to it (including brokers and dealers in securities).

The Trustee may act as principal or agent in the making or disposing of any investment. Upon the Written Request of the Authority, the Trustee shall sell or present for redemption any Permitted Investments so purchased whenever it shall be necessary to provide moneys to meet any required payment, transfer, withdrawal or disbursement from the fund to which such Permitted Investments is credited, and the Trustee shall not be liable or responsible for any loss resulting from

any investment made or sold pursuant to this Section. For purposes of investment, the Trustee may commingle moneys in any of the funds and accounts established hereunder.

The Trustee may make any investments hereunder through the bond or investment department or trust investment department of the entity acting as Trustee hereunder, or those of such entity's parent or any affiliate, and such entity, or its parent or affiliate, as applicable, shall be entitled to its normal, customary and reasonable compensation for such services.

The entity acting as Trustee hereunder, or any of its affiliates, may act as sponsor, advisor or manager in connection with any investments made by the Trustee hereunder and such entity, or its affiliate, as applicable, shall be entitled to its normal, customary and reasonable compensation for such services.

The Authority and the City acknowledge that, to the extent regulations of the Comptroller of the Currency or other applicable regulatory entity grant the Authority and the City the right to receive brokerage confirmations of security transactions as they occur, at no additional cost, the Authority and the City specifically waive receipt of such confirmations to the extent permitted by law.

# 5.08 [Claims Upon the Insurance Policy and Payments by and to the Insurer.]

5.09 Payments by the Insurer as a Result of Nonpayment. The Insurer shall be entitled to pay principal or interest on the Series BG Bonds that shall become Due for Payment but shall be unpaid by reason of Nonpayment by the Authority (as such terms are defined in the Insurance Policy) and any amounts due on the Series BG Bonds as a result of acceleration of the maturity thereof in accordance with this Indenture, whether or not the Insurer has received a Notice of Nonpayment (as such terms are defined in the Insurance Policy) or a claim upon the Insurance Policy.

### **ARTICLE VI**

#### **COVENANTS**

- **6.01** Compliance with Agreements. The Trustee will not authenticate or deliver any Bonds in any manner other than in accordance with the provisions hereof, and the Authority and the City will not suffer or permit any default by them to occur hereunder, but will faithfully comply with, keep, observe and perform all the agreements, conditions, covenants and terms hereof required to be complied with, kept, observed and performed by them.
- **6.02** Compliance with Ground Lease and Lease Agreement. The Authority and the City will faithfully comply with, keep, observe and perform all the agreements, conditions, covenants and terms contained in the Ground Lease and the Lease Agreement required to be complied with, kept, observed and performed by them and, together with the Trustee, will enforce the Ground Lease and the Lease Agreement against the other party thereto in accordance with their respective terms.
- 6.03 Observance of Laws and Regulations. The Authority, the City and the Trustee will faithfully comply with, keep, observe and perform all valid and lawful obligations or regulations now or hereafter imposed on them by contract, or prescribed by any law of the United States of America or of the State of California, or by any officer, board or commission having jurisdiction or control, as a condition of the continued enjoyment of each and every franchise, right or privilege now owned or hereafter acquired by them, including their right to exist and carry on their respective businesses, to

the end that such franchises, rights and privileges shall be maintained and preserved and shall not become abandoned, forfeited or in any manner impaired.

6.04 Other Liens. The City will keep the Property and all parts thereof free from judgments and materialmen's and mechanics' liens and free from all claims, demands, encumbrances and other liens of whatever nature or character, and free from any claim or liability which materially impairs the City in conducting its business or utilizing the Property, and the Trustee at its option (after first giving the City ten days' written notice to comply therewith and failure of the City to so comply within such ten-day period) may, but is in no event obligated to, defend against any and all actions or proceedings, or may pay or compromise any claim or demand asserted in any such actions or proceedings; provided, however, that, in defending against any such actions or proceedings or in paying or compromising any such claims or demands, the Trustee shall not in any event be deemed to have waived or released the City from liability for or on account of any of its agreements and covenants contained herein, or from its liability hereunder and to perform such agreements and covenants.

So long as any Bonds are Outstanding, none of the Trustee, the Authority or the City shall create or suffer to be created any pledge of or lien the amounts on deposit in any of the funds or accounts created hereunder, other than the pledge and lien hereof.

The Authority, the City and the Trustee shall not encumber the Property other than in accordance with the Ground Lease, the Lease Agreement, the Indenture and the Assignment Agreement.

- **6.05 Prosecution and Defense of Suits**. The City will promptly, upon request of the Trustee (which request the Trustee is not required to make), take such action from time to time as may be necessary or proper to remedy or cure any cloud upon or defect in the title to the Property or any part thereof, whether now existing or hereafter developing, will prosecute all actions, suits or other proceedings as may be appropriate for such purpose and will indemnify and save the Trustee harmless from all cost, damage, expense or loss, including attorneys' fees and expenses, which it or the Owners may incur by reason of any such cloud, defect, action, suit or other proceeding.
- **6.06** Accounting Records and Statements. The Trustee will keep proper accounting records in which complete and correct entries shall be made of all transactions relating to the receipt, deposit and disbursement of the Base Rental Payments, and such accounting records shall be available for inspection by the Authority and the City at reasonable hours and under reasonable conditions.
- **6.07 Recordation and Filing**. The City will record, or cause to be recorded, with the appropriate county recorder, the Lease Agreement, the Ground Lease and the Assignment Agreement, or memoranda thereof.

#### **6.08** [Reserved].

**6.09 Continuing Disclosure**. The City will comply with and carry out all of the provisions of the Continuing Disclosure Certificate. Notwithstanding any other provision of this Indenture, failure of the City to comply with the Continuing Disclosure Certificate shall not constitute an event of default hereunder; provided, however, that the Trustee may (and, at the written direction of any Participating Underwriter or the holders of at least 25% of the aggregate principal

amount of Outstanding Series BG Bonds, and upon being indemnified to its reasonable satisfaction therefor, shall) or any holder or beneficial owner of the Series BG Bonds may take such actions as may be necessary and appropriate to compel performance, including seeking mandate or specific performance by court order.

**6.10 Further Assurances**. Whenever and so often as requested to do so by the Trustee, the Authority and the City will promptly execute and deliver or cause to be executed and delivered all such other and further assurances, documents or instruments and promptly do or cause to be done all such other and further things as may be necessary or reasonably required in order to further and more fully vest in the Trustee all advantages, benefits, interests, powers, privileges and rights conferred or intended to be conferred upon it hereby or by the Assignment Agreement, the Ground Lease or the Lease Agreement.

#### ARTICLE VII

#### **DEFAULT AND LIMITATIONS OF LIABILITY**

- **7.01** Action on Default. If an event of default (within the meaning of Article VI of the Lease Agreement) shall happen, then such event of default shall constitute an event of default hereunder. The Trustee shall give notice, as assignee of the Authority, of an event of default under the Lease Agreement to the City. In each and every case during the continuance of an event of default, the Trustee may and, at the direction of the Owners of not less than a majority of the aggregate principal amount of Bonds then Outstanding, and upon being indemnified to its reasonable satisfaction therefor, shall, upon notice in writing to the City and the Authority, exercise any of the remedies granted to the Authority under the Lease Agreement and, in addition, take whatever action at law or in equity may appear necessary or desirable to enforce its rights as assignee pursuant to the Assignment Agreement or to protect and enforce any of the rights vested in the Trustee or the Owners by this Indenture or by the Bonds, either at law or in equity or in bankruptcy or otherwise, whether for the specific enforcement of any covenant or agreement or for the enforcement of any other legal or equitable right, including any one or more of the remedies set forth in Section 7.02 hereof.
- **7.02** Other Remedies of the Trustee. Subject to the provisions of Section 7.01 hereof, the Trustee shall have the right:
- (a) by mandamus or other action or proceeding or suit at law or in equity to enforce its rights against the Authority or the City or any member, director, officer or employee thereof, and to compel the Authority or the City or any such member, director, officer or employee to perform or carry out its or his or her duties under law and the agreements and covenants required to be performed by it or him or her contained herein;
- (b) by suit in equity to enjoin any acts or things which are unlawful or violate the rights of the Trustee; or
- (c) by suit in equity upon the happening of any event of default hereunder to require the Authority and the City to account as the trustee of an express trust.

Nothing herein shall be deemed to authorize the Trustee to authorize or consent to or accept or adopt on behalf of any Owner any plan of reorganization, arrangement, adjustment or composition

affecting the Bonds or the rights of any Owner thereof, or to authorize the Trustee to vote in respect of the claim of any Owner in any such proceeding without the approval of the Owners so affected.

7.03 Non-Waiver. A waiver of any default or breach of duty or contract by the Trustee shall not affect any subsequent default or breach of duty or contract or impair any rights or remedies on any such subsequent default or breach of duty or contract. No delay or omission by the Trustee to exercise any right or remedy accruing upon any default or breach of duty or contract shall impair any such right or remedy or shall be construed to be a waiver of any such default or breach of duty or contract or an acquiescence therein, and every right or remedy conferred upon the Trustee by law or by this Article may be enforced and exercised from time to time and as often the Trustee shall deem expedient.

If any action, proceeding or suit to enforce any right or to exercise any remedy is abandoned or determined adversely to the Trustee or any Owner, then subject to any adverse determination, the Trustee, such Owner, the Authority and the City shall be restored to their former positions, rights and remedies as if such action, proceeding or suit had not been brought or taken.

- **7.04** Remedies Not Exclusive. Subject to the provisions of Section 7.01 hereof, no remedy herein conferred upon or reserved to the Trustee is intended to be exclusive of any other remedy, and each such remedy shall be cumulative and shall be in addition to every other remedy given hereunder or now or hereafter existing in law or in equity or by statute or otherwise and may be exercised without exhausting and without regard to any other remedy conferred by any law. The assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other appropriate right or remedy.
- **7.05 No Liability by the Authority to the Owners**. Except as expressly provided herein, the Authority shall not have any obligation or liability to the Owners with respect to the payment when due of the Base Rental Payments by the City, or with respect to the performance by the City of the other agreements and covenants required to be performed by it contained in the Lease Agreement or herein, or with respect to the performance by the Trustee of any right or obligation required to be performed by it contained herein.
- 7.06 No Liability by the City to the Owners. Except for the payment when due of the Base Rental Payments and the performance of the other agreements and covenants required to be performed by it contained in the Lease Agreement, the Ground Lease or herein, the City shall not have any obligation or liability to the Owners with respect to this Indenture or the preparation, execution, delivery or transfer of the Bonds or the disbursement of the Base Rental Payments by the Trustee to the Owners, or with respect to the performance by the Trustee of any right or obligation required to be performed by it contained herein.
- **7.07 No Liability of the Trustee to the Owners**. Except as expressly provided herein, the Trustee shall not have any obligation or liability to the Owners with respect to the payment when due of the Base Rental Payments by the City, or with respect to the performance by the Authority or the City of the other agreements and covenants required to be performed by them contained in the Lease Agreement, the Ground Lease or herein.
- **7.08** Application of Amounts After Default. All payments received by the Trustee with respect to the rental of the Property after a default by the City pursuant to Article VI of the Lease Agreement (including, without limitation, any proceeds received in connection with the sale,

assignment or sublease of the Authority's right, title and interest in the Ground Lease), and all damages or other payments received by the Trustee for the enforcement of any rights and powers of the Trustee under Article VI of the Lease Agreement, shall be deposited into the Base Rental Payment Fund and as soon as practicable thereafter applied, together with all other funds held hereunder (except funds in the Rebate Fund):

- (a) to the payment of all amounts due the Trustee under Article VIII hereof;
- (b) to the payment of all amounts then due for interest on the Bonds, in respect of which, or for the benefit of which, money has been collected (other than Bonds which have become payable-prior to such event of default and money for the payment of which is held by the Trustee), ratably without preference or priority of any kind, according to the amounts of interest on such Bonds due and payable;
- (c) to the payment of all amounts then due for principal of the Bonds, in respect of which, or for the benefit of which, money has been collected (other than Bonds which have become payable prior to such event of default and money for the payment of which is held by the Trustee), ratably without preference or priority of any kind, according to the amounts of principal of such Bonds due and payable; and
- (d) to the extent Policy Costs or Insurer Reimbursement Amounts are due and payable to the Insurer, to the payment of such amounts.
- 7.09 Trustee May Enforce Claims Without Possession of Bonds. All rights of action and claims under this Indenture or the Bonds may be prosecuted and enforced by the Trustee without the possession of any of the Bonds or the production thereof in any proceeding relating thereto, and any such proceeding instituted by the Trustee shall be brought in its own name as trustee of an express trust, and any recovery of judgment shall, after provision for the payment of the reasonable compensation, expenses, disbursements and advances of the Trustee, its agents and counsel, be for the ratable benefit of the Owners in respect of which such judgment has been recovered.
- Limitation on Suits. No Owner of any Bond shall have any right to institute any proceeding, judicial or otherwise, with respect to this Indenture, or for the appointment of a receiver or Trustee, or for any other remedy hereunder, unless (a) such Owner shall have previously given written notice to the Trustee of a continuing event of default, (b) the Owners of not less than 25% of the aggregate principal amount of Bonds then Outstanding shall have made written request to the Trustee to institute proceedings in respect of such event of default in its own name as Trustee hereunder, (c) such Owner or Owners shall have afforded to the Trustee indemnity reasonably satisfactory to the Trustee against the costs, expenses and liabilities to be incurred in compliance with such request, (d) the Trustee for 60 days after its receipt of such notice, request and offer of indemnity shall have failed to institute any such proceedings, and (e) no direction inconsistent with such written request shall have been given to the Trustee during such 60 day period by the Owners of a majority of the aggregate principal amount of Bonds then Outstanding; it being understood and intended that no one or more Owners shall have any right in any manner whatever by virtue of, or by availing of, any provision of this Indenture to affect, disturb or prejudice the rights of any other Owner, or to obtain or seek to obtain priority or preference over any other Owner or to enforce any right under this Indenture, except in the manner herein provided and for the equal and ratable benefit of all the Owners.

**Insurer Rights.** The Insurer shall be deemed to be the sole holder of the Series BG Bonds for the purpose of exercising any voting right or privilege or giving any consent or direction or taking any other action that the Series BG Bond Owners are entitled to take pursuant to the Indenture pertaining to (i) defaults and remedies and (ii) the duties and obligations of the Trustee. In furtherance thereof and as a term of the Indenture and each Series BG Bond, the Trustee (solely with respect to the Series BG Bonds) and each Series BG Bond Owner appoint the Insurer as their agent and attorney-in-fact and agree that the Insurer may at any time during the continuation of any proceeding by or against the Authority under the United States Bankruptcy Code or any other applicable bankruptcy, insolvency, receivership, rehabilitation or similar law (an "Insolvency Proceeding") direct all matters relating to such Insolvency Proceeding, including without limitation, (A) all matters relating to any claim or enforcement proceeding in connection with an Insolvency Proceeding (a "Claim"), (B) the direction of any appeal of any order relating to any Claim, (C) the posting of any surety, supersedeas or performance bond pending any such appeal, and (D) the right to vote to accept or reject any plan of adjustment. In addition, the Trustee (solely with respect to the Series BG Bonds) and each Series BG Bond Owner delegate and assign to the Insurer, to the fullest extent permitted by law, the rights of the Trustee and each Series BG Bond Owner in the conduct of any Insolvency Proceeding, including, without limitation, all rights of any party to an adversary proceeding or action with respect to any court order issued in connection with any such Insolvency Proceeding. Remedies granted to the Owners shall expressly include mandamus.

#### **ARTICLE VIII**

### THE TRUSTEE

8.01 Employment of the Trustee. The Authority hereby appoints and employs the Trustee to receive, deposit and disburse the Base Rental Payments, to authenticate, deliver and transfer the Bonds and to perform the other functions contained herein, all in the manner provided herein and subject to the conditions and terms hereof. By executing and delivering this Indenture, the Trustee accepts the appointment and employment hereinabove referred to and accepts the rights and obligations of the Trustee provided herein, subject to the conditions and terms hereof. Other than when an event of default has occurred and is continuing, the Trustee undertakes to perform such duties and only such duties as are specifically set forth in this Indenture, and no implied covenants or obligations shall be read into this Indenture against the Trustee. In case an event of default has occurred and is continuing, the Trustee shall exercise such of the rights and powers vested in it by this Indenture, and use the same degree of care and skill in their exercise, as a prudent man would exercise or use under the circumstances in the conduct of his own affairs. The Trustee hereby covenants and agrees that it will not encumber the Property.

8.02 Duties, Removal and Resignation of the Trustee. The Authority may, by an instrument in writing, remove the Trustee initially a party hereto and any successor thereto unless an event of default shall have occurred and then be continuing, and shall remove the Trustee initially a party hereto and any successor thereto if at any time (a) requested to do so by an instrument or concurrent instruments in writing signed by the Owners of a majority of the aggregate principal amount of Bonds at the time Outstanding (or their attorneys duly authorized in writing), or (b) the Trustee shall cease to be eligible in accordance with the following sentence, and shall appoint a successor Trustee. The Trustee and any successor Trustee shall be a banking corporation or association or trust company having (or if such banking corporation or association or trust company is a member of a bank holding company, its bank holding company has) a combined capital (exclusive of borrowed capital) and surplus of at least \$50,000,000 and subject to supervision or

examination by federal or state authorities. If such banking corporation or association or trust company publishes a report of condition at least annually, pursuant to law or to the requirements of any supervising or examining authority above referred to, then for the purposes of this Section the combined capital and surplus of such bank or trust company shall be deemed to be its combined capital and surplus as set forth in its most recent report of condition so published.

The Trustee may at any time resign by giving written notice of such resignation to the Authority and the City and by giving notice, by first class mail, postage prepaid, of such resignation to the Owners at their addresses appearing on the Registration Books. Upon receiving such notice of resignation, the Authority shall promptly appoint a successor Trustee by an instrument in writing; provided, however, that in the event the Authority does not appoint a successor Trustee within 30 days following receipt of such notice of resignation, the resigning Trustee may, at the expense of the Authority, petition the appropriate court having jurisdiction to appoint a successor Trustee. Any resignation or removal of a Trustee and appointment of a successor Trustee shall become effective only upon acceptance of appointment by the successor Trustee.

Any corporation, association or agency into which the Trustee may be converted or merged, or with which it may be consolidated, or to which it may sell or transfer its corporate trust business and assets as a whole or substantially as a whole, or any corporation or association resulting from any such conversion, sale, merger, consolidation or transfer to which it is a party, provided that such entity meets the combined capital and surplus requirements of this Section, ipso facto, shall be and become successor trustee under this Indenture and vested with all the trusts, powers, discretions, immunities, privileges and all other matters as was its predecessor, without the execution or filing of any instrument or any further act, deed or conveyance on the part of any of the parties hereto, anything herein to the contrary notwithstanding.

**8.03** Compensation of the Trustee. The City shall from time to time, subject to any written agreement then in effect with the Trustee, pay the Trustee reasonable compensation for all its services rendered hereunder and reimburse the Trustee for all its reasonable advances and expenditures (which shall not include "overhead expenses" except as such expenses are included as a component of the Trustee's stated annual fees) hereunder, including but not limited to advances to and reasonable fees and reasonable expenses of accountants, agents, appraisers, consultants or other experts, and counsel not directly employed by the Trustee but an attorney or firm of attorneys retained by the Trustee, employed by it in the exercise and performance of its rights and obligations hereunder. The Trustee may take whatever legal actions are lawfully available to it directly against the Authority or the City.

The City shall, to the extent permitted by law, indemnify and save the Trustee harmless against any liabilities, costs, claims or expenses, including those of its attorneys, which it may incur in the exercise and performance of its powers and duties hereunder, under the Lease Agreement, or in connection with any document or transaction contemplated hereunder or thereunder, including the enforcement of any remedies and the defense of any suit, and which are not due to its negligence or its misconduct. The duty of the City to indemnify the Trustee shall survive the termination and discharge of this Indenture and the earlier removal or resignation of the Trustee.

No provision of this Indenture shall require the Trustee to expend or risk its own funds or otherwise incur any financial liability in the performance of any of its duties hereunder or in the exercise of any of its rights or powers hereunder.

Upon an Event of Default, and only upon an Event of Default, the Trustee shall have a first lien with right of payment prior to payment on account of principal of and premium, if any, and interest on any Bond, upon the trust estate for the foregoing fees, charges and expenses incurred by it. When the Trustee incurs expenses or renders services after the occurrence of an Event of Default, such expenses and the compensation for such services are intended to constitute expenses of administration under any federal or state bankruptcy, insolvency, arrangement, moratorium, reorganization or other debtor relief law.

**8.04 Protection of the Trustee**. The Trustee shall be protected and shall incur no liability in acting or proceeding in good faith upon any affidavit, bond, certificate, consent, notice, request, requisition, resolution, statement, telegram, voucher, waiver or other paper or document which it shall in good faith believe to be genuine and to have been adopted, executed or delivered by the proper party or pursuant to any of the provisions hereof, and the Trustee shall be under no duty to make any investigation or inquiry as to any statements contained or matters referred to in any such instrument, but may accept and rely upon the same as conclusive evidence of the truth and accuracy of such statements. The Trustee shall be under no obligation to exercise any of the rights or powers vested in it by this Indenture at the request or direction of any of the Owners of the Bonds pursuant to this Indenture, unless such Owners shall have offered to the Trustee security or indemnity, reasonably satisfactory to the Trustee, against the reasonable costs, expenses and liabilities which might be incurred by it in compliance with such request or direction. The Trustee may consult with counsel, who may be counsel to the Authority or the City, with regard to legal questions, and the opinion of such counsel shall be full and complete authorization and protection in respect to any action taken or suffered by it hereunder in good faith in accordance therewith.

The Trustee shall not be responsible for the sufficiency of the Bonds or the Lease Agreement, or of the assignment made to it by the Assignment Agreement, or for statements made in any preliminary or final official statement relating to the Bonds, or of the title to the Property.

Whenever in the administration of its rights and obligations hereunder the Trustee shall deem it necessary or desirable that a matter be proved or established prior to taking or suffering any action hereunder, such matter (unless other evidence in respect thereof be herein specifically prescribed) may be deemed to be conclusively proved and established by a Written Certificate of the City or a Written Certificate of the Authority, and such certificate shall be full warrant to the Trustee for any action taken or suffered under the provisions hereof upon the faith thereof, but in its discretion the Trustee may, in lieu thereof, accept other evidence of such matter or may require such additional evidence as it deems reasonable.

The Trustee may buy, sell, own, hold and deal in any of the Bonds and may join in any action which any Owner may be entitled to take with like effect as if the Trustee were not a party hereto, The Trustee, either as principal or agent, may also engage in or be interested in any financial or other transaction with the Authority or the City, and may act as agent, depository or trustee for any committee or body of Owners or of owners of obligations of the Authority or the City as freely as if it were not the Trustee hereunder.

The Trustee may, to the extent reasonably necessary, execute any of the trusts or powers hereof and perform any rights and obligations required of it hereunder by or through agents, attorneys or receivers, and shall be entitled to advice of counsel concerning all matters of trust and its rights and obligations hereunder, and the Trustee shall not be answerable for the negligence or misconduct of any such agent, attorney or receiver selected by it with reasonable care; provided,

however, that in the event of any negligence or misconduct of any such attorney, agent or receiver, the Trustee shall in a commercially reasonable manner pursue all remedies of the Trustee against such agent, attorney or receiver. The Trustee shall not be liable for any error of judgment made by it in good faith unless it shall be proved that the Trustee was negligent in ascertaining the pertinent facts.

The Trustee shall not be answerable for the exercise of any trusts or powers hereunder or for anything whatsoever in connection with the funds established hereunder, except only for its own willful misconduct, negligence or breach of an obligation hereunder.

The Trustee shall not be deemed to have knowledge of an event of default unless it has actual knowledge thereof.

The Trustee may, on behalf of the Owners, intervene in any judicial proceeding to which the Authority or the City is a party and which, in the opinion of the Trustee and its counsel, affects the Bonds or the security therefor, and shall do so if requested in writing by the Owners of at least 5% of the aggregate principal amount of Bonds then Outstanding, provided the Trustee shall have no duty to take such action unless it has been indemnified to its reasonable satisfaction against all risk or liability arising from such action.

The Trustee's rights to immunities and protection from liability hereunder and its rights to payment of its fees and expenses shall survive its resignation or removal and final payment or defeasance of the Bonds.

All indemnifications and releases from liability granted herein to the Trustee shall extend to the directors, officers, employees and agents of the Trustee (including its counsel).

The permissive right of the Trustee to do things enumerated in this Indenture shall not be construed as a duty, and the Trustee shall not be answerable for other than its negligence or willful default. The Trustee shall have no responsibility or liability with respect to any information, statements or recitals in any offering memorandum or other disclosure material prepared or distributed with respect to the issuance of these Bonds. The Trustee shall not be accountable for the use or application by the Borrower of any of the Bonds or the proceeds thereof or for the use or application of any money paid over by the Trustee in accordance with the provisions of this Indenture or for the use and application of money received by any paying agent.

The Trustee shall not be liable to the parties hereto or deemed in breach or default hereunder if and to the extent its performance hereunder is prevented by reason of force majeure. The term "force majeure" means an occurrence that is beyond the control of the Trustee and could not have been avoided by exercising due care. Force majeure shall include but not limited to acts of God, terrorism, war, riots, strikes, fire, floods, earthquakes, epidemics or other similar occurrences.

The Trustee agrees to accept and act upon instructions or directions pursuant to this Indenture sent by unsecured e-mail, facsimile transmission or other similar unsecured electronic methods; provided, however, that the Trustee shall have received an incumbency certificate listing persons designated to give such instructions or directions and containing specimen signatures of such designated persons, which such incumbency certificate shall be amended and replaced whenever a person is to be added or deleted from the listing. If the Authority or the City elects to give the Trustee e-mail or facsimile instructions (or instructions by a similar electronic method) and the

Trustee in its discretion elects to act upon such instructions, the Trustee's understanding of such instructions shall be deemed controlling. The Trustee shall not be liable for any losses, costs or expenses arising directly or indirectly from the Trustee's reliance upon and compliance with such instructions notwithstanding such instructions conflict or are inconsistent with a subsequent written instruction. The Authority and the City agree to assume all risks arising out of the use of such electronic methods to submit instructions and directions to the Trustee, including, without limitation, the risk of the Trustee acting on unauthorized instructions, and the risk of interception and misuse by third parties.

In acting or omitting to act pursuant to the Lease Agreement or Ground Lease, the Trustee shall be entitled to all of the rights, immunities and indemnities accorded to it under this Indenture and the Lease Agreement, including, but not limited to, this Article VIII.

**8.05 Notice to Insurer by Trustee**. The Trustee shall notify the Insurer of any failure of the Authority to provide notices, certificates and other information under the transaction documents of which the Trustee has actual or deemed knowledge pursuant to Section 8.04 hereof.

#### **ARTICLE IX**

#### MODIFICATION OR AMENDMENTS

#### 9.01 Modifications and Amendments Permitted.

- This Indenture and the rights and obligations of the Authority, the City, the Owners of the Bonds and the Trustee may be modified or amended from time to time and at any time by a Supplemental Indenture, which the Authority, the City and the Trustee may enter into with the written consent of the Owners of a majority of the Owners of the Bonds then Outstanding and the written consent of the Insurer, so long as the Insurer is not in default on any obligation under the Insurance Policy, which shall have been filed with the Trustee. No such modification or amendment shall (i) extend the fixed maturity of any Bonds, or reduce the amount of principal thereof or the rate of interest thereon, or extend the time of payment, without the consent of the Owner of each Bond so affected, or (ii) eliminate the aforesaid consent of the Insurer or reduce the aforesaid percentage of Bonds the consent of the Owners of which is required to effect any such modification or amendment, or (iii) permit the creation of any lien on the Base Rental Payments and other assets pledged under this Indenture prior to or on a parity with the lien created by this Indenture or deprive the Owners of the Bonds of the lien created by this Indenture on such Base Rental Payments and other assets (except as expressly provided in this Indenture), without the consent of the Owners of all of the Bonds then Outstanding. It shall not be necessary for the consent of the Bond Owners to approve the particular form of any Supplemental Indenture, but it shall be sufficient if such consent shall approve the substance thereof.
- (b) This Indenture and the rights and obligations of the Authority, the City, the Trustee, and the Owners of the Bonds may also be modified or amended from time to time and at any time by a Supplemental Indenture, which the Authority, the City and the Trustee may enter into without the consent of any Bond Owners but with the prior written consent of the Insurer for any one or more of the following purposes:
- (i) to add to the covenants and agreements of the Authority or the City in this Indenture contained other covenants and agreements thereafter to be observed, to pledge or

assign additional security for the Bonds (or any portion thereof), or to surrender any right or power herein reserved to or conferred upon the Authority or the City;

- (ii) to make such provisions for the purpose of curing any ambiguity, inconsistency or omission, or of curing or correcting any defective provision contained in this Indenture;
- (iii) to provide for the issuance of one or more Series of Additional Bonds, and to provide the terms and conditions under which such Series of Additional Bonds may be issued, subject to and in accordance with the provisions of Article III hereof;
- (iv) to modify, amend or supplement this Indenture in such manner as to permit the qualification hereof under the Trust Indenture Act of 1939, as amended, or any similar federal statute hereafter in effect, and to add such other terms, conditions and provisions as may be permitted by said act or similar federal statute;
- (v) to modify, amend or supplement this Indenture in such manner as to cause interest on any series of tax-exempt Bonds to be excludable from gross income for purposes of federal income taxation by the United States of America; and
- (vi) in any other respect whatsoever as the Authority and the City may deem necessary or desirable, provided that such modification or amendment does not materially adversely affect the interests of the Bond Owners hereunder, in the opinion of Bond Counsel filed with the Authority, the City and the Trustee.
- (c) Promptly after the execution by the Authority, the City and the Trustee of any Supplemental Indenture, the Trustee shall mail a notice (the form of which shall be furnished to the Trustee by the Authority), by first class mail postage prepaid, setting forth in general terms the substance of such Supplemental Indenture, to the Owners of the Bonds at the respective addresses shown on the Registration Books. Any failure to give such notice, or any defect therein, shall not, however, in any way impair or affect the validity of any such Supplemental Indenture.
- (d) No Supplemental Indenture shall modify any of the rights or obligations of the Trustee without its prior written consent.
- **9.02 Effect of Supplemental Indenture**. Upon the execution of any Supplemental Indenture pursuant to this Article, this Indenture shall be deemed to be modified and amended in accordance therewith, and the respective rights, duties and obligations under this Indenture of the Authority, the City, the Trustee and all Owners of Bonds Outstanding shall thereafter be determined, exercised and enforced hereunder subject in all respects to such modification and amendment, and all the terms and conditions of any such Supplemental Indenture shall be deemed to be part of the terms and conditions of this Indenture for any and all purposes.
- **9.03** Endorsement of Bonds; Preparation of New Bonds. Bonds delivered after the execution of any Supplemental Indenture pursuant to this Article may, and if the Authority so determines shall, bear a notation by endorsement or otherwise in form approved by the Authority and the Trustee as to any modification or amendment provided for in such Supplemental Indenture, and, in that case, upon demand of the Owner of any Bonds Outstanding at the time of such execution and presentation of his Bonds for the purpose at the Office of the Trustee a suitable notation shall be

made on such Bonds. If the Supplemental Indenture shall so provide, new Bonds so modified as to conform, in the opinion of the Authority and the Trustee, to any modification or amendment contained in such Supplemental Indenture, shall be prepared and executed by the Authority and authenticated by the Trustee, and upon demand of the Owners of any Bonds then Outstanding shall be exchanged at the Office of the Trustee, without cost to any Bond Owner, for Bonds then Outstanding, upon surrender for cancellation of such Bonds, in equal aggregate principal amount of the same interest rate and maturity.

- **9.04** Amendment of Particular Bonds. The provisions of this Article shall not prevent any Bond Owner from accepting any amendment as to the particular Bonds held by such Owner.
- **9.05** Effect of Insurance Policy. In determining whether any amendment, consent, waiver or other action to be taken, or any failure to take action, under the Indenture would adversely affect the security for the Series BG Bonds or the rights of the Owners, the Trustee shall consider the effect of any such amendment, consent, waiver, action or inaction as if there were no Insurance Policy.

#### **ARTICLE X**

#### **DEFEASANCE**

10.01 Discharge of Indenture. If the Authority shall pay or cause to be paid or there shall otherwise be paid to the Owners of all Outstanding Bonds the principal thereof and the interest and premium, if any, thereon at the times and in the manner stipulated herein and therein, then the Owners of such Bonds shall cease to be entitled to the pledge of the Base Rental Payments and the other assets as provided herein, and all agreements, covenants and other obligations of the Authority and the City to the Owners of such Bonds hereunder shall thereupon cease, terminate and become void and be discharged and satisfied. In such event, the Trustee shall execute and deliver to the Authority and the City all such instruments as may be necessary or desirable to evidence such discharge and satisfaction, and the Trustee shall pay over or deliver to the City all money or securities held by it pursuant hereto which are not required for the payment of the principal of and interest and premium, if any, on such Bonds,

Subject to the provisions of the above paragraph, when any of the Bonds shall have been paid and if, at the time of such payment, the Authority and the City shall have kept, performed and observed all of the covenants and promises in such Bonds and in this Indenture required or contemplated to be kept, performed and observed by them on or prior to that time, then this Indenture shall be considered to have been discharged in respect of such Bonds and such Bonds shall cease to be entitled to the lien of this Indenture and such lien and all covenants, agreements and other obligations of the Authority and the City hereunder shall cease, terminate become void and be completely discharged as to such Bonds.

Notwithstanding the satisfaction and discharge of this Indenture or the discharge of this Indenture in respect of any Bonds, those provisions of this Indenture relating to the maturity of the Bonds, interest payments and dates thereof, exchange and transfer of Bonds, replacement of mutilated, destroyed, lost or stolen Bonds, the safekeeping and cancellation of Bonds, non-presentment of Bonds, and the duties of the Trustee in connection with all of the foregoing, shall remain in effect and shall be binding upon the Trustee and the Owners of the Bonds and the Trustee shall continue to be obligated to hold in trust any moneys or investments then held by the Trustee for the payment of the principal of and interest and premium, if any, on the Bonds, to pay to the Owners

of Bonds the funds so held by the Trustee as and when such payment becomes due. Notwithstanding the satisfaction and discharge of this Indenture or the discharge of this Indenture in respect of any Bonds, those provisions of this Indenture relating to the compensation and indemnity of the Trustee shall remain in effect and shall be binding upon the Trustee, the City and the Authority.

10.02 Bonds Deemed To Have Been Paid. If moneys shall have been set aside and held by the Trustee for the payment or redemption of any Bonds and the interest thereon at the maturity or redemption date thereof, such Bonds shall be deemed to have been paid within the meaning and with the effect provided in Section 10.01 hereof. Any Outstanding Bonds shall prior to the maturity date or redemption date thereof be deemed to have been paid within the meaning of and with the effect expressed in Section 10.01 hereof if (a) in case any of such Bonds are to be redeemed on any date prior to their maturity date, the Authority shall have given to the Trustee in form satisfactory to it irrevocable instructions to mail, on a date in accordance with the provisions of Section 4.02 hereof, notice of redemption of such Bonds on said redemption date, said notice to be given in accordance with Section 4.02 hereof, (b) there shall have been deposited with the Trustee either (i) money in an amount which shall be sufficient, or (ii) Federal Securities that are not subject to redemption other than at the option of the holder thereof, the interest on and principal of which when paid will provide money which, together with the money, if any deposited with the Trustee at the same time, shall, as verified by an independent certified public accountant in a report (acceptable to the Insurer) filed with the Authority, the City, the Insurer and the Trustee, be sufficient to pay when due the interest to become due on such Bonds on and prior to the maturity date or redemption date thereof, as the case may be, and the principal of and premium, if any, on such Bonds, (c) the Authority shall have delivered or cause to be delivered an (i) escrow agreement (which shall be acceptable in form and substance to the Insurer, so long as the Insurer is not in default on any obligation under the Insurance Policy); and (ii) an opinion of Bond Counsel addressed to the Authority, the City, the Trustee and the Insurer to the effect that such Series BG Bonds have been discharged in accordance with the Indenture (which opinion may rely upon and assume the accuracy of the independent certified public accountant report referred to above) and (d) in the event such Bonds are not by their terms subject to redemption within the next succeeding 60 days, the Authority shall have given the Trustee in form satisfactory to it irrevocable instructions to mail as soon as practicable, a notice to the owners of such Bonds that the deposit required by clause (b) above has been made with the Trustee and that such Bonds, are deemed to have been paid in accordance with this Section and stating the maturity date or redemption date upon which money is to be available for the payment of the principal of and premium, if any, on such Bonds. The Insurer shall be provided with final drafts of the documents set forth in Section 10.02(c) hereof not less than five Business Days prior to the funding of the escrow.

This Indenture shall not be discharged with respect to the Series BG Bonds until all Policy Costs and Insurer Reimbursement Amounts and any other amounts due to the Insurer payable pursuant to Section 11.22 hereof have been paid. The obligation to pay such amounts shall expressly survive the payment in full of the Series BG Bonds.

The Series BG Bonds shall be deemed Outstanding under this Indenture unless and until they are in fact paid and retired or the above criteria are met.

10.03 Payment of Bonds After Discharge of Indenture. Notwithstanding any provisions of this Indenture, to the extent permitted by law, any moneys held by the Trustee in trust for the payment of the principal of, or premium or interest on, any Bonds and remaining unclaimed for two years after the date of deposit of such moneys, shall be repaid to the Authority (without liability for interest) free from the trusts created by this Indenture, and all liability of the Trustee with respect to

such moneys shall thereupon cease; provided, however, that before the repayment of such moneys to the Authority as aforesaid, the Trustee may (at the cost of the Authority) first mail, by first class mail postage prepaid, to the Owners of Bonds which have not yet been paid, at the respective addresses shown on the Registration Books, a notice, in such form as may be deemed appropriate by the Trustee with respect to the Bonds so payable and not presented and with respect to the provisions relating to the repayment to the Authority of the moneys held for the payment thereof.

#### **ARTICLE XI**

#### **MISCELLANEOUS**

- 11.01 Benefits of Indenture Limited to Parties. Nothing contained herein, expressed or implied, is intended to give to any person other than the Authority, the City, the Trustee, the Insurer and the Owners any claim, remedy or right under or pursuant hereto, and any agreement, condition, covenant or term required herein to be observed or performed by or on behalf of the Authority or the City shall be for the sole and exclusive benefit of the Trustee, the Insurer and the Owners.
- 11.02 Successor Deemed Included in all References to Predecessor. Whenever the Authority, the City or the Trustee, or any officer thereof, is named or referred to herein, such reference shall be deemed to include the successor to the powers, duties and functions that are presently vested in the Authority, the City or the Trustee, or such officer, and all agreements, conditions, covenants and terms required hereby to be observed or performed by or on behalf of the Authority, the City or the Trustee, or any officer thereof, shall bind and inure to the benefit of the respective successors thereof whether so expressed or not.
- 11.03 Execution of Documents by Owners. Any declaration, request or other instrument which is permitted or required herein to be executed by Owners may be in one or more instruments of similar tenor and may be executed by Owners in person or by their attorneys appointed in writing. The fact and date of the execution by any Owner or his attorney of any declaration, request or other instrument or of any writing appointing such attorney may be proved by the certificate of any notary public or other officer authorized to take acknowledgments of deeds to be recorded in the state or territory in which he purports to act that the person signing such declaration, request or other instrument or writing acknowledged to him the execution thereof, or by an affidavit of a witness of such execution duly sworn to before such notary public or other officer, or by such other proof as the Trustee may accept which it may deem sufficient.

The ownership of any Bonds and the amount, payment date, number and date of owning the same may be proved by the Registration Books.

Any declaration, request or other instrument in writing of the Owner of any Bond shall bind all future Owners of such Bond with respect to anything done or suffered to be done by the Authority, the City or the Trustee in good faith and in accordance therewith.

11.04 Waiver of Personal Liability. Notwithstanding anything contained herein to the contrary, no member, officer or employee of the Authority or the City shall be individually or personally liable for the payment of any moneys, including without limitation, the principal of or interest on the Bonds, but nothing contained herein shall relieve any member, officer or employee of the City or the Authority from the performance of any official duty provided by any applicable provisions of law, by the Lease Agreement or hereby.

- 11.05 **Destruction of Bonds**. Whenever in this Indenture provision is made for the cancellation by the Trustee and the delivery to the Authority of any Bonds, the Trustee may, in lieu of such cancellation and delivery, destroy such Bonds.
- 11.06 Funds and Accounts. Any fund or account required to be established and maintained herein by the Trustee may be established and maintained in the accounting records of the Trustee either as an account or a fund, and may, for the purposes of such accounting records, any audits thereof and any reports or statements with respect thereto, be treated either as an account or a fund, but all such records with respect to all such funds and accounts shall at an times be maintained in accordance with sound accounting practice and with due regard for the protection of the security of the Bonds and the rights of the Owners.

The Trustee may commingle any of the moneys held by it hereunder for investment purposes only; provided, however, that the Trustee shall account separately for the moneys in each fund or account established pursuant to this Indenture. The Trustee may establish such funds and accounts as it deems necessary or appropriate to perform its obligations hereunder.

- 11.07 Article and Section Headings Gender and References. The singular form of any word used herein, including the terms defined in Section 1.01 hereof, shall include the plural, and vice versa, unless the context otherwise requires. The use herein of a pronoun of any gender shall include correlative words of the other genders. The headings or titles of the several Articles and Sections hereof and the table of contents appended hereto shall be solely for convenience of reference and shall not affect the meaning, construction or effect hereof. All references herein to "Articles," "Sections," subsections or clauses are to the corresponding Articles, Sections, subsections or clauses hereof, and the words "hereby," "herein," "hereof," "hereto," "herewith," "hereunder" and other words of similar import refer to this Indenture as a whole and not to any particular Article, Section, subsection or clause thereof.
- 11.08 Partial Invalidity. If any one or more of the agreements, conditions, covenants or terms required herein to be observed or performed by or on the part of the Authority, the City or the Trustee shall be contrary to law, then such agreement or agreements, such condition or conditions, such covenant or covenants or such term or terms shall be null and void to the extent contrary to law and shall be deemed separable from the remaining agreements, conditions, covenants and terms hereof and shall in no way affect the validity hereof or of the Bonds, and the Owners shall retain all the benefit, protection and security afforded to them under any applicable provisions of law. The Authority, the City and the Trustee hereby declare that they would have executed this Indenture, and each and every Article, Section, paragraph, subsection, sentence, clause and phrase hereof and would have authorized the execution and delivery of the Bonds pursuant hereto irrespective of the fact that any one or more Articles, Sections, paragraphs, subsections, sentences, clauses or phrases hereof or the application thereof to any person or circumstance may be held to be unconstitutional, unenforceable or invalid.
- 11.09 Disqualified Bonds. In determining whether the Owners of the requisite aggregate principal amount of Bonds have concurred in any demand, request, direction, consent or waiver under this Indenture, Bonds which are actually known by the Trustee to be owned or held by or for the account of the Authority or the City, or by any Person directly or indirectly controlling or controlled by, or under direct or indirect common control with, the Authority or the City, shall be disregarded and deemed not to be Outstanding for the purpose of any such determination; except that, in determining whether the Trustee shall be protected in relying upon any such demand, request,

direction, consent or waiver of an Owner, only Bonds which the Trustee actually knows to be owned or held by or for the account of the Authority or the City, or by any Person directly or indirectly controlling or controlled by, or under direct or indirect common control with, the Authority or the City, shall be disregarded unless all Bonds are so owned or held, in which case such Bonds shall be considered Outstanding for the purpose of such determination. Bonds so owned which have been pledged in good faith may be regarded as Outstanding for the purposes of this Section if the pledgee shall establish to the satisfaction of the Trustee the pledgee's right to vote such Bonds and that the pledgee is not a Person directly or indirectly controlling or controlled by, or under direct or indirect common control with, the Authority or the City. In case of a dispute as to such right, any decision by the Trustee taken upon the advice of counsel shall be full protection to the Trustee.

- 11.10 Money Held for Particular Bonds. The money held by the Trustee for the payment of the interest, principal or premium due on any date with respect to particular Bonds (or portions of Bonds in the case of Bonds redeemed in part only) shall, on and after such date and pending such payment, be set aside on its books and held in trust by it for the Owners of the Bonds entitled thereto, subject, however, to the provisions of Section 10.03 hereof but without any liability for interest thereon.
- 11.11 Payment on Non-Business Days. In the event any payment is required to be made hereunder on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day with the same effect as if made on such non-Business Day.
- 11.12 California Law. This Indenture shall be construed and governed in accordance with the laws of the State of California.
- 11.13 Notices. All written notices to be given hereunder shall be given by mail to the party entitled thereto at its address set forth below, or at such other address as such party may provide to the other parties in writing from time to time, namely: [Update]

City of Pomona

505 South Garey Avenue
Pomona, California 91769
Attention: City Manager

If to the Authority:

Pomona Public Financing Authority
c/o City of Pomona
505 South Garey Avenue
Pomona, California 91769
Attention: Chairperson

If to the Trustee:

Zions Bank, a division of ZB, N.A.,

If to the City:

Attention:	
Notices to the Insurer shall be provided to _	

Each such notice, statement, demand, consent, approval, authorization, offer, designation, request or other communication hereunder shall be deemed delivered to the party to whom it is

- addressed (a) if personally served or delivered, upon delivery, (b) if given by electronic communication, whether by telex, telegram or telecopier, upon the sender's receipt of an appropriate answer back or other written acknowledgment, (c) if given by registered or certified mail, return receipt requested, deposited with the United States mail postage prepaid, 72 hours after such notice is deposited with the United States mail, (d) if given by overnight courier, with courier charges prepaid, 24 hours after delivery to said overnight courier, or (d) if given by any other means, upon delivery at the address specified in this Section.
- 11.14 Notice to Rating Agencies. The Trustee shall provide S&P, if the Bonds are then rated by S&P, and Moody's, if the Bonds are then rated by Moody's, with prompt notice of any substitution or release of property pursuant to Section 9.03 of the Lease Agreement.
- 11.15 Execution in Counterparts. This Indenture may be executed in several counterparts, each of which shall be deemed an original, and all of which shall constitute but one and the same instrument.
- 11.16 Insurer as Third Party Beneficiary. The Insurer is intended as a third party beneficiary to this Indenture.
- 11.17 Impairment of Insurer's Rights. No contract shall be entered into or any action taken by which the rights of the Insurer to reimbursement from Rental Payments or the Property may be impaired or prejudiced in any material respect except upon obtaining the prior written consent of the Insurer.
- 11.18 Insurer Consideration. The rights granted to the Insurer under the Indenture to request, consent to or direct any action are rights granted to the Insurer in consideration of its issuance of the Insurance Policy. Any exercise by the Insurer of such rights is merely an exercise of the Insurer's contractual rights and shall not be construed or deemed to be taken for the benefit, or on behalf, of the Series BG Bonds Owners and such action does not evidence any position of the Insurer, affirmative or negative, as to whether the consent of the Series BG Bonds Owners or any other person is required in addition to the consent of the Insurer.
- 11.19 Amounts Paid by Insurer. Amounts paid by the Insurer under the Insurance Policy and the Series BG Reserve Surety Policy shall not be deemed paid for purposes of this Indenture and the Series BG Bonds relating to such payments shall remain Outstanding and continue to be due and owing until paid by the City and the Authority in accordance with this Indenture. This Indenture shall not be discharged unless all amounts due or to become due to the Insurer have been paid in full or duly provided for.
- 11.20 Covenant to Preserve Priority. The Authority covenants and agrees to take such action (including, as applicable, filing of Uniform Commercial Code financing statements and continuations thereof) as is necessary from time to time to preserve the priority of the pledge of the Base Rental Payments set forth in Section 5.01 under applicable law.
- 11.21 Subrogation and Survival of Obligations. The Insurer shall, to the extent it makes any payment of principal of or interest on the Series BG Bonds, become subrogated to the rights of the recipients of such payments in accordance with the terms of the Insurance Policy (which subrogation rights shall also include the rights of any such recipients in connection with any Insolvency Proceeding). Each obligation of the Authority and/or the City to the Insurer under this

Indenture or any supplement thereto or amendment thereof shall survive discharge or termination of this Indenture or any supplement thereto or amendment thereof.

- 11.22 Reimbursement of Fees. The Authority and/or the City shall pay or reimburse the Insurer any and all charges, fees, costs and expenses that the Insurer may reasonably pay or incur in connection with (a) the administration, enforcement, defense or preservation of any rights or security in this Indenture, the Lease, the Ground Lease or any supplement thereto or amendment thereof, (b) the pursuit of any remedies under this Indenture, the Lease, the Ground Lease or any supplement thereto or amendment thereof or otherwise afforded by law or equity, (c) any amendment, waiver or other action with respect to, or related to, this Indenture, the Lease, the Ground Lease or any supplement thereto or amendment thereof whether or not executed or completed, or (d) any litigation or other dispute in connection with this Indenture, the Lease, the Ground Lease or any supplement thereto or amendment thereof or the transactions contemplated thereby, other than costs resulting from the failure of the Insurer to honor its obligations under the Insurance Policy. The Insurer reserves the right to charge a reasonable fee as a condition to executing any amendment, waiver or consent proposed in respect of this Indenture, the Lease, the Ground Lease or any supplement thereto or amendment thereof.
- 11.23 Provision of Information. So long as the Insurer is not in default on any obligation under the Insurance Policy and such default is not continuing, the Insurer shall be provided with the following information by the Authority, the City or Trustee, as the case may be:
- (a) Annual audited financial statements within 225 days after the end of the City's Fiscal Year (together with a certification of the City that it is not aware of any default or event of default under this Indenture), and the City's annual budget within 30 days after the approval thereof together with such other information, data or reports as the Insurer shall reasonably request from time to time;
- (b) Notice of any draw upon the Reserve Fund within two Business Days after knowledge thereof other than (i) withdrawals of amounts in excess of the Reserve Requirement and (ii) withdrawals in connection with a refunding of the Series BG Bonds;
- (c) Notice of any event of default known to the Trustee, the Authority or City within five Business Days after knowledge thereof;
- (d) Prior notice of the advance refunding or redemption of any of the Series BG Bonds, including the principal amount, maturities and CUSIP numbers thereof;
- (e) Notice of the resignation or removal of the Trustee and the appointment of, and acceptance of duties by, any successor thereto;
- (f) Notice of the commencement of any proceeding by or against the Authority or the City commenced under the United States Bankruptcy Code or any other applicable bankruptcy, insolvency, receivership, rehabilitation or similar law (an "Insolvency Proceeding");
- (g) Notice of the making of any claim in connection with any Insolvency Proceeding seeking the avoidance as a preferential transfer of any payment of principal of, or interest on, the Series BG Bonds:

- (h) A full original transcript of all proceedings relating to the execution of any amendment, supplement, or waiver to this Indenture or any supplement thereto or amendment thereof;
- (i) All reports, notices and correspondence to be delivered to Owners under the terms of this Indenture and any supplement thereto or amendment thereof; and
- (j) All information furnished pursuant to the Continuing Disclosure Certificate executed by the City with respect to the Series BG Bonds shall also be provided to the Insurer simultaneously with the furnishing of such information.
- **11.24 Additional Information**. The Insurer shall have the right to receive such additional information as it may reasonably request.
- 11.25 Discussion of and Access to Information. The Authority and the City will permit the Insurer to discuss the affairs, finances and accounts of the Authority and the City or any information the Insurer may reasonably request regarding the security for the Series BG Bonds with appropriate officers of the Authority and the City and will use commercially reasonable efforts to enable the Insurer to have access to the facilities, books and records of the Authority and the City on any Business Day upon reasonable prior notice.

**IN WITNESS WHEREOF**, the Authority and the City have caused this Indenture to be signed in their respective names by their representative thereunto duly authorized, and the Trustee, in token of its acceptance of the trusts created hereunder, has caused this Indenture to be signed in its corporate name by its officer thereunto duly authorized, all as of the day and year first above written,

# 

City Clerk

ZIONS BANK, a division of ZB, N.A., as Trustee	
By:	
Authorized Officer	

#### **EXHIBIT A**

#### FORM OF SERIES BG BOND

No		\$

UNLESS THIS BOND IS PRESENTED BY AN AUTHORIZED REPRESENTATIVE OF THE DEPOSITORY TRUST COMPANY TO THE TRUSTEE FOR REGISTRATION OF TRANSFER, EXCHANGE OR PAYMENT, AND ANY BOND ISSUED IS REGISTERED IN THE NAME OF CEDE & CO. OR SUCH OTHER NAME AS REQUESTED BY AN AUTHORIZED REPRESENTATIVE OF THE DEPOSITORY TRUST COMPANY AND ANY PAYMENT IS MADE TO CEDE & CO., ANY TRANSFER, PLEDGE OR OTHER USE HEREOF FOR VALUE OR OTHERWISE BY OR TO ANY PERSON IS WRONGFUL SINCE THE REGISTERED OWNER HEREOF, CEDE & CO., HAS AN INTEREST HEREIN.

## POMONA PUBLIC FINANCING AUTHORITY 2017 LEASE REVENUE REFUNDING BONDS, SERIES BG (FEDERALLY TAXABLE)

INTEREST RATE%	MATURITY DATE July 1, 20	<b>DATED DATE</b> , 2017	CUSIP —
REGISTERED OWNE	R: CEDE & CO.		
PRINCIPAL AMOUNT	٠.	DOLLARS	

The Pomona Public Financing Authority (the "Authority"), for value received, hereby promises to pay, solely from the Base Rental Payments (as hereinafter defined) or amounts in certain funds and accounts held under the Indenture (as hereinafter defined), to the Registered Owner identified above or registered assigns (the "Registered Owner"); on the Maturity Date identified above or on any earlier redemption date, the Principal Amount identified above in lawful money of the United States of America; and to pay interest thereon at the Rate of Interest identified above in like lawful money from the date hereof payable semiannually on July 1 and January 1 in each year, commencing [July 1, 2017] (the "Interest Payment Dates"), until payment of such Principal Amount in full. This Bond shall bear interest from the Interest Payment Date next preceding the date of authentication of this Bond (unless this Bond is authenticated on or before an Interest Payment Date and after the close of business on the fifteenth calendar day of the month next preceding such Interest Payment Date, whether or not such day is a Business Day, in which event it shall bear interest from such Interest Payment Date, or unless this Bond is authenticated on or prior to June 15, 2017, in which event it shall bear interest from the Dated Date identified above; provided, however, that if, at the time of authentication of this Bond, interest is in default on this Bond, interest on this Bond shall be payable from the date to which interest hereon has been paid in full, payable on each Interest Payment Date). The Principal Amount hereof is payable upon surrender hereof upon maturity or earlier redemption at the Office of the Trustee (as hereinafter defined). Interest hereon is payable by wire or check of Zions Bank, a division of ZB, N.A., as Trustee (the "Trustee"), mailed by first class mail, postage prepaid, on each Interest Payment Date to the Registered Owner hereof at the address

of the Registered Owner shown on the Registration Books at the close of business on the fifteenth calendar day of the month next preceding such Interest Payment Date. "Office of the Trustee" means the principal corporate trust office of the Trustee in Costa Mesa, California, or such other office as may be specified to the Authority and the City of Pomona (the "City") by the Trustee in writing, except that with respect to presentation of Bonds for payment or for registration of transfer and exchange such term shall mean the office or the agency of the Trustee at which, at any particular time, its corporate trust agency shall be conducted as specified to the Authority and the City by the Trustee in writing.

This Bond is one of a series of a duly authorized issue of bonds issued for the purpose of defeasing the Pomona Public Financing Authority's outstanding 2006 Lease Revenue Bonds, Series AU and Series AV and a portion of the City's outstanding Pension Obligation Refunding Bonds (Federally Taxable) Series 2006 AR, and is one of the series of bonds designated "Pomona Public Financing Authority 2017 Lease Revenue Refunding Bonds, Series BG (Federally Taxable)" (the "Series BG Bonds") in the aggregate principal amount of \$\_\_\_\_\_\_\_\_. The Series BG Bonds are issued pursuant to the Indenture, dated as of \_\_\_\_\_\_\_\_\_\_\_ 1, 2017 (the "Indenture"), by and among the Authority, the City and the Trustee, and this reference incorporates the Indenture herein, and by acceptance hereof the owner of this Bond assents to said terms and conditions. Pursuant to and as more particularly provided in the Indenture, additional bonds ("Additional Bonds"), may be issued by the Authority secured by a lien on a parity with the lien securing the Series BG Bonds. The Series BG Bonds and any Additional Bonds are collectively referred to as the "Bonds." The Indenture is entered into, and this Bond is issued under, the Marks-Roos Local Bond Pooling Act of 1985 (the "Act") and the laws of the State of California.

The Series BG Bonds are authorized to be issued in the form of fully registered bonds without coupons in denominations of \$5,000 or any integral multiple thereof ("Authorized Denominations").

The Series BG Bonds shall be subject to redemption, in whole or in part, on any date, in Authorized Denominations, from and to the extent of any net insurance proceeds received with respect to all or a portion of the Property leased under the Lease Agreement, remaining after payment therefrom of all reasonable expenses incurred in the collection thereof, deposited by the Trustee in the Redemption Fund established under the Indenture, at a Redemption Price equal to the principal amount of the Series BG Bonds to be redeemed, plus accrued interest thereon to the date of redemption, without premium.

The Series BG Bonds maturing on or after July 1, 20\_\_, shall be subject to optional redemption, in whole or in part, on any date on or after July 1, 20\_\_, in Authorized Denominations, from and to the extent of prepaid Base Rental Payments paid pursuant the Lease Agreement, at a Redemption Price equal to the principal amount of the Series BG Bonds to be redeemed, plus accrued interest thereon to the date of redemption.

The Series BG Bonds maturing on July 1, 20\_\_ are subject to mandatory sinking fund redemption in part (by lot) on each July 1 on and after July 1, 20\_\_, in integral multiples of \$5,000 at a Redemption Price of the principal amount thereof plus accrued interest to the date fixed for redemption, without premium, in accordance with the following schedule:

Sinking Fund Redemption Date (July 1)

Principal Amount
To Be Redeemed

(maturity)

The Series BG Bonds maturing on July 1, 20\_\_ are subject to mandatory sinking fund redemption in part (by lot) on each July 1 on and after July 1, 20\_\_, in integral multiples of \$5,000 at a Redemption Price of the principal amount thereof plus accrued interest to the date fixed for redemption, without premium, in accordance with the following schedule:

Sinking Fund Redemption Date (July 1)

Principal Amount To Be Redeemed

(maturity)

The Trustee on behalf and at the expense of the Authority shall mail (by first class mail) notice of any redemption to the respective owners of any Series BG Bonds designated for redemption, at their respective addresses appearing on the Registration Books, at least 20 but not more than 60 days prior to the date fixed for redemption; provided, however, that neither failure to receive any such notice so mailed, nor any defect therein, shall affect the validity of the proceedings for the redemption of such Series BG Bonds or the cessation of accrual of interest thereon from and after the date fixed for redemption. The Redemption Price of the Series BG Bonds to be redeemed shall be paid only upon presentation and surrender thereof at the Office of the Trustee. From and after the date fixed for redemption of any Series BG Bonds, interest on such Series BG Bonds will cease to accrue and become payable.

Subject to the limitations and upon payment of the charges, if any, provided in the Indenture, fully registered Series BG Bonds may be exchanged at the Office of the Trustee for a like aggregate principal amount and maturity of fully registered Series BG Bonds of other authorized denominations.

This Bond is transferable by the Registered Owner hereof, in person or by his duly authorized attorney, but only in the manner, subject to the limitations and upon payment of the charges provided in the Indenture, and upon surrender and cancellation of this Bond. Upon such transfer a new fully registered Series BG Bond or Series BG Bonds, in Authorized Denominations, for the same aggregate principal amount will be issued to the transferee in exchange herefor. The Authority, the City and the Trustee may treat the Registered Owner hereof as the absolute owner hereof for all purposes, and the Authority, the City and the Trustee shall not be affected by any notice to the contrary.

The Indenture and the rights and obligations of the Authority, the City, the owners of the Bonds, the Insurer and the Trustee may be modified or amended from time to time and at any time in the manner, to the extent, and upon the terms provided in the Indenture; provided that no such modification or amendment shall (a) extend the fixed maturity of any Bonds, or reduce the principal thereof or the rate of interest thereon, or extend the time of payment, without the consent of the owner of each Bond so affected, or, (b) eliminate the consent of the Insurer as set forth in the Indenture or reduce the aforesaid percentage of Bonds the consent of the Owners of which is required to effect any such modification or amendment, or (c) permit the creation of any lien on the Base Rental Payments and other assets pledged under the Indenture prior to or on a parity with the lien created by the Indenture or deprive the owners of the Bonds of the lien created by the Indenture on such the Base Rental Payments and such other assets (except as expressly provided in the Indenture), without the consent of the owners of all Bonds then outstanding.

The Indenture contains provisions permitting the Authority to make provision for the payment of interest on, and the principal and premium, if any, of any of the Bond so that such Bonds shall no longer be deemed to be outstanding under the terms of the Indenture.

All obligations of the Authority under the Indenture shall be special obligations of the Authority, payable solely from Rental Payments and the other assets pledged therefor under the Indenture; provided, however, that all obligations of the Authority under the Bonds shall be special obligations of the Authority, payable solely from Base Rental Payments and the other assets pledged therefor under the Indenture. Neither the faith and credit nor the taxing power of the Authority, the City or the State of California, or any political subdivision thereof, is pledged to the payment of the Bonds.

**IN WITNESS WHEREOF**, the Authority has caused this Bond to be signed in its name and on its behalf by the facsimile signatures of its Chair and Secretary, all as of the Dated Date identified above.

POMONA PUBLIC FINANCING AUTHORITY

	By:		
	•	Chairperson	
Attest:			
Secretary			

# [FORM OF TRUSTEE'S CERTIFICATE OF AUTHENTICATION]

described in the within-mentioned Indenture and
ZIONS BANK, a division of ZB, N.A., as Trustee
By:Authorized Signatory

## STATEMENT OF INFORMATION

# [FORM OF ASSIGNMENT]

For value, received the undersigne	ed hereby sells, assigns and transfers unto whose address and social security or other tax
	the within-mentioned Bond and hereby attorney, to
transfer the same on the registration books of premises.	the Trustee with full power of substitution in the
Dated:	
Signature Guaranteed:	
Note: Signature(s) must be guaranteed by an eligible guarantor.	Note: The signature(s) on this Assignment must correspond with the name(s) as written on the face of the within pond in every particular without alteration or enlargement or any change whatsoever